

# **MASTERARBEIT / MASTER'S THESIS**

# Titel der Masterarbeit / Title of the Master's Thesis "Vowel reduction in Standard Southern British:

# a cognitive grammar approach"

verfasst von / submitted by Matthias Mittendorfer, BA BA

angestrebter akademischer Grad / in partial fulfilment of the requirements for the degree of Master of Arts (MA)

Wien, 2020 / Vienna 2020

Studienkennzahl It. Studienblatt / degree programme code as it appears on the student record sheet:

Studienrichtung It. Studienblatt / degree programme as it appears on the student record sheet:

Betreut von / Supervisor:

UA 066 812

English Language and Linguistics

Univ.-Prof. Mag. Dr. Nikolaus Ritt

#### **ACKNOWLEDGEMENTS**

I am indebted to a number of people without whom it would not have been possible to finish this thesis. First and foremost, I would like to express my gratitude Univ.-Prof. Mag. Dr. Nikolaus Ritt for generously accepting my somewhat exotic thesis topic. Even though not working in the field of Cognitive Grammar, he was always happy to discuss issues and problems arising during the research process.

I would also like to thank my parents, who have always supported me no matter what. They never pressured me into finishing my studies and thus allowed me to make the most of my time at university. Moreover, I wish to extend my thanks to all my friends and colleagues who have patiently endured my constant complaining. Without all of you, this would not have been possible.

	List of abbreviations	i
	List of figures	i
	List of tables	i
1	INTRODUCTION	1
1.		••••••
2.	COGNITIVE GRAMMAR AND ITS USAGE-BASED NATURE	4
		_
	2.1. Principles of usage-based approaches	4
		4
	2.1.1.1 ne content requirement	4
	2.1.2. The status of cognition, categorisation and the role of experience	
	2.1.5. The fulle/list fallacy	0
	2.2. Grammar as a network of categories	7
		,
2	COCNITIVE CDAMMAD AND DHONOLOCY	0
з.	CUGNITIVE GRAWIWAR AND PHONOLOGY	
	2.1. Same basis about logical concents in CC	0
	3.1. Some basic phonological concepts in CG	9
	3.1.2 Phonological features in CG	9 11
	5.1.2.1 hohological realarcs in CG	
	3.2. Key concepts in CG phonology	13
	3.2.1. First- and second-order schemas.	
	3.2.2. Schema interaction and well-formedness principles	14
	3.2.3. Some parallels to Optimality Theory	
4.	THE SOUND SYSTEM OF STANDARD SOUTHERN BRITISH	19
	4.1. RP and Standard Southern British	19
	4.2. The sound system of SSB	20
	4.2.1.Vowels	20
	4.2.2.Consonants	
5.	METHODOLOGICAL ASPECTS	23
	5.1. Dictionary and transcription symbols	23
	J I J	
	5.2. Type of data	24
6	VOWEL REDUCTION IN SSR	25
0.	VOWEL REDUCTION IN SSB	
	6.1 Strong and vowal reduction	75
	U.1. SUESS AND VOWEI FEURCHON	23
	6.2 Data and discussion	77
	6.2.1 Short vowels	27
	···	······································

# TABLE OF CONTENTS

	6.2.2.Long monophthongs	
	6.2.3.Diphthongs	
	6.2.4.Reduction in SSB: a brief summary	41
7.	A CG ANALYSIS OF VOWEL REDUCTION IN SSB	42
	7.1. Reduction of short vowels	42
	7 1 1 The schemas	42
	7.1.2.Analysis	46
	7.2. Reduction of long monophthongs	55
	7.2.1. The schemas	
	7.2.2.Analysis	56
	7.3. Reduction of diphthongs	58
	7.3.1.The schemas	
	7.3.2.Analysis	60
	7.4. Exceptions: Frequency effects in CG	65
	7.4.1.Non-reduction of low frequency items	
	7.4.2.Reduction of high frequency items	68
8.	VOWEL REDUCTION IN RUSSIAN: A BRIEF SKETCH	
	8.1. Contemporary Standard Russian	71
	8.2. The Russian sound system	71
	8.2.1. Vowel system	71
	8.2.2.Consonant system	73
	8.3. Vowel reduction in Russian: the data	73
	8.3.4. Analysis of Russian vowel reduction	75
	8.3.5. The schemas	
	8.3.6.Analysis	
9.	CONCLUSION	80
10.	. REFERENCES	
11	APPENDIX A	89
11,		
12.	. APPENDIX B	96
	12.1.English abstract	96
	12.2.German abstract	97

# List of abbreviations

V	schematic short vowel
VV	schematic long vowel or diphthong
С	schematic consonant
C <sub>SON</sub>	sonorant
σ	syllable
\$	syllable boundary
φ	phonological foot
()	foot boundary

# List of figures

Figure 1 Grammar as a network (based on Nesset (2008: 12))	7
Figure 2 The phoneme as a complex category (based on Nesset's account of Russian vow	els
(2008: 32))	10
Figure 3 First and second-order schemas exemplified	13
Figure 4 Schema interaction exemplified	15
Figure 5 Anti-clockwise vowel shift in SSB (based on Lindsey 2019: 18)	21
Figure 6 Syllable schemas	43
Figure 7 Reduction schemas for short vowels	45
Figure 8 Foot-structure in English	
Figure 9 Reduction schemas for long vowels	56
Figure 10 Reduction schemas for diphthongs	59
Figure 11 The vowel system in Russian (based on Akišina & Baranovskaya 1980: 86)	72
Figure 12 Reduction schemas for Russian	76

# List of tables

Table 1 Diphthongs in SSB	22
Table 2 Transcription system for vowels in SSB (based on Lindsey 2019: 146)	24
Table 3 Reduction of short vowels	27
Table 4 Reduction of long monophthongs	32
Table 5 Reduction of diphthongs	37
Table 6 Reduction patterns in SSB summarised	41
Table 7 The Russian consonant inventory (Akišina & Baranovskaya 1980: 57)	73
Table 8 Vowel reduction in Russian	74
Table 9 Data on short vowels	89
Table 10 Data on long monophthongs	91
Table 11 Data on diphthongs	93

#### 1. Introduction

The sound structure of a language is no less a 'cognitive' phenomenon than semantic structure. Taylor (2002:79)

Cognitive linguistics, an approach typically concerned with semantic phenomena, has long relegated phonology to its periphery. In fact, on average fewer than one article per year is published on phonological issues in one of the major journals of the field, viz. Cognitive *Linguistics* (Nathan 2015: 253). Naturally, this raises the question as to why phonology takes such a marginal role within an otherwise quite successful research paradigm. Taylor (2002: 79) suggests that one of the reasons for the rather tentative treatment of phonology stems from "a widespread misconception about the scope of Cognitive Linguistics in general" (2002: 79), which essentially equates the term cognitive with conceptual and conceptual with semantic. Even some remarks by Langacker (1987: 12) seem to support the view that meaning is central: "From the symbolic nature of language follows the centrality of meaning to virtually all linguistic concerns. Meaning is what language is all about" (1987: 12). This seemingly strong emphasis on semantics, in combination with the fact that most of the work within Cognitive Grammar (henceforth CG) deals with semantic phenomena, appears to reinforce the view that phonology is simply not of importance in the framework. However, excluding phonology from the scope of CG raises several issues troublesome to the theory on the whole (cf. Taylor 2002: 79f.). A comprehensive linguistic theory needs to be able to account for all aspects of language and not restrict itself to semantics only. If it fails in doing so, it "is only half a theory of language" (2002: 79). Furthermore, phonology is as much a cognitive phenomenon as any other aspect of linguistic structure. Thus, phonological units (e.g. feet, syllables or phonemes) are conceptual in so far, as they can be understood as concepts in the minds of speakers (2002: 80).

Phonology in CG takes a radically different perspective on grammar than the traditionally dominant Generative Grammar and its offsprings. While the present thesis will not put forward a detailed comparison of the two frameworks, a brief comparison of the two approaches may be appropriate at this point. The primary benefit of adopting a usage-based theory such as CG lies in the central role of phonotactics (Kumashiro 2000: 1). In other words, constraints are captured using cognitive schemas that are abstracted from actually occurring expressions. In contrast, such a theoretical restriction does not exist in Generative

Phonology (and Optimality Theory; cf. Prince & Smolensky 2004). Rather, Kumashiro (2000: 1) argues that in such approaches "constrains/rules can be posited arbitrarily and independently of actually-occurring expressions" (2000: 1). Conversely, as a non-reductionist and maximalist framework, CG takes as its starting point expressions found in language use, from which generalisations, i.e. schemas, are abstracted bottom-up (cf. section 2.1.1). Conversely, Generative Grammar is reductionist in nature (2000: 2). Surface forms are understood to be the product of underlying forms to which certain rules and constraints have been applied top-down. In contrast, these constraints and rules are typically put forward without any consideration of actually occurring expressions (2000: 2). CG, taking a usage-based approach (cf. section 2.1), provides a principled manner of establishing constraints as schemas (based on actual utterances) and thus does not have to fall back on arbitrary decisions made by the researcher.

Couched in a CG framework, the present thesis will provide an analysis of vowel reduction in British English. More specifically, it will explore ways in which generally known cognitive mechanisms, i.e. schema formation and categorising relationships, used in the traditional areas of cognitive grammar, i.e. semantics, can explain unstressed vowel reduction in Standard Southern British (SSB). The version of CG adopted in this thesis assumes that language users can "employ [alternative strategies] when activating schemas in categorization networks" (Nesset 2008: 14). These strategies will be modelled in the proposed analysis. Moreover, it will be shown that reduction processes in Russian can be accounted for by the same theoretical constructs developed for English and that no arbitrary mechanisms are needed to cover for the data in Russian. Unstressed vowel reduction is a pervasive phenomenon influencing the development of a variety of languages. Consequently, it has been investigated from various theoretical standpoints, such as, for example, generative phonology (Chomsky & Halle 1991 [1968]), Optimality Theory (Crosswhite 2001) or Government Phonology (Pöchtrager 2018). Even though studied extensively in various languages (especially in Romance and Slavic; cf. Barnes 2006), the term vowel reduction is often used to describe an array of different linguistic phenomena (Crosswhite 2001: 3). Extreme definitions range from "the wholesale deletion of unstressed vowels" (2001: 3) to "non-neutralizing changes in the pronunciation of both stressed and unstressed vowels" [original emphasis] (2001: 3). In this thesis, vowel reduction is defined as the

neutralisation of "two or more [...] vowel qualities [...] in a stress-dependent fashion" (2001: 3).

Relatively little has been published on the topic so far with respect to British English. Most discussions of unstressed vowel reduction either take an American English perspective (e.g. Flemming & Johnson 2007) or seem to be unconcerned about the variety used (Burzio 1994, Chomsky & Halle 1991 [1968], Crosswhite 2001 and others).<sup>1</sup> Moreover, due to the rather limited body of work on phonological phenomena within Cognitive Linguistics in general and CG in particular, accounts of vowel reduction within the framework are practically absent in the relevant literature (with the exception of Nesset 2006 on Russian). Thus, this thesis will prove useful in several ways. It will not only contribute to the rather scarce literature in the field, but also provide a cognitively plausible account of the phenomenon within cognitive grammar. A framework of phonology within cognitive grammar is necessary in so far, as a theory of language such as cognitive linguistics needs to be able to deal with all aspects of language. As pointed out by Taylor (2002: 79), "the exclusion of phonology would mean that Cognitive Grammar could not lay claim to being a comprehensive theory of language" (2002: 79).

The remainder of this thesis is organised as follows. Section 2 will outline the core assumptions of cognitive grammar, viz. its usage-based approach to the study of language. In section 3, the present thesis will explore how CG can be applied to phonology. In order to do so, some basic phonological concepts and how they may be accounted for in CG will be discussed. Following this, the foundations of the theory used in the analysis of vowel reduction will be examined. Section 4 will outline the variety of English chosen for the analysis. Standard Southern British deviates from classical RP in several important ways, which need to be defined prior to the analysis. Section 5 will introduce some of the methodological aspects underlying the present study. This is followed by a detailed exploration of the data set in section 6. Each set of vowels, viz. short vowels, long monophthongs and diphthongs, will be discussed on the basis of selected examples (see appendix for the full data set). Having considered all the relevant aspects of the data set, the thesis will move on to the analysis proposed in this thesis in section 7. The final section of this thesis, section 8, presents an analysis of vowel reduction in Contemporary Standard

<sup>&</sup>lt;sup>1</sup> Even though it can reasonably be assumed that they take some American variety as their reference.

Russian (CSR). It will be shown that the theoretical assumptions used in the analysis of English reduction can successfully be applied to Russian as well.

#### 2. Cognitive Grammar and its usage-based nature

This section will consider some fundamental assumptions made in CG. A number of principles important to usage-based models in general will be explored in section 2.1. Section 2.2 will build on this foundation and outline how grammar can be understood as a network of categories.

# 2.1. Principles of usage-based approaches

#### 2.1.1. The content requirement

As a usage-based model, CG puts its central emphasis on language use. This is particularly evident in the content requirement, postulated by Langacker (1987: 53): "The only structures permitted in the grammar of a language [...] are (1) phonological, semantic or symbolic structures that actually occur in linguistic expressions; (2) schemas for such structures; and (3) categorizing relationships involving the elements in (1) and (2)" (1987: 53). It follows that CG does not allow any underlying representations or empty elements lacking both phonological and semantic content. Nor does it allow any theoretical tools "valid only in a particular subdomain of linguistics (e.g. phonology, syntax)" (Kumashiro 2000: 11). Consequently, any analysis of linguistic phenomena couched in a CG framework needs to employ the same (general) cognitive mechanisms allowed by the content requirement, i.e. schemas and categorising relationships (cf. section 2.2). The content requirement has important implications as to how the grammar of a language is conceptualised. At this point, it suffices to briefly return to the "maximalist', 'non-reductive' [and] 'bottomup" (Langacker 1999: 91) nature of the approach (cf. section 2.1.3). It is maximalist in so far as every conventional linguistic unit such as, for example, any word, is assumed to be contained in the mental grammar of a speaker. Moreover, as a non-reductive approach, CG does not reduce surface variety to a set of underlying representations but takes "the entire inventory of actually-occurring expressions [as the starting point]" (Kumashiro 2000: 13), from which constraints and generalisations are drawn in a bottom-up fashion. The content requirement stated above is in agreement with the larger usage-based model developed. Bybee (2001: 6-8) proposes a number of principles of such approaches of which some match or advance CG's content requirement. These will be discussed in the following.

#### 2.1.2. The status of cognition, categorisation and the role of experience

An important aspect of usage-based approaches is the assumption that the human linguistic faculty "is part of the more general human cognitive faculty" (Kumashiro 2000: 7). This is to say that there is, in principle, no difference in properties between linguistic and non-linguistic mental representations (Bybee 2001: 7). Put differently, for cognitive linguistics, "linguistic cognition simply is cognition" (Janda 2015: 132). A psychological phenomenon crucial in cognition in general and in language use in particular is, for instance, *categorisation*. According to Bybee (2001: 7), categorisation is heavily grounded in similarity or identity. It should be emphasised again that there are no theoretical mechanisms which are only relevant to particular subdomains of linguistics. Consequently, categorisation not only plays a crucial role in semantic phenomena, but also in phonology. For instance, it can be assumed that categorisation regulates "the storage of phonological percepts" (2001: 7). Thus, two different tokens of the same phoneme (i.e. allophones) may be subsumed under the same category. This is to say that speakers form categories over sets of sounds based on their experience of language. As categorisation is substantial in the analytical part of this thesis, it will be taken up again in some detail in section 2.2 below.<sup>2</sup>

Usage-based models place special emphasis on the role of linguistic experience, which affects the way language is represented in the speaker's memory. Such approaches view grammar as "the cognitive organization of one's experience with language" (Bybee 2006: 711). As a result of the continuous exposure to language, grammar is a dynamic system constantly shaped anew (Dąbrowska 2004: 213). In other words, the structure of language changes or remains constant because of the way it is used by speakers (Bybee 2001: 5-6). This view is supported by Langacker (2010: 109), who claims that "[linguistic] structure emerges from usage, is immanent in usage, and is influenced by usage on an ongoing basis" (2010: 109). What follows from the focus on experience is that frequency is taken to have a strong effect on how a linguistic unit is represented in the speaker's memory (Bybee 2001, 2006; Dąbrowska 2004). The more frequent a particular linguistic unit is, i.e. the more often it is accessed, the stronger its representation in the mental grammar becomes and the more easily it is activated (Dąbrowska 2004: 213). While Bybee (2001: 6-7) does not use any specific term for this psychological phenomenon, Langacker (2000: 3) refers to the effect

<sup>&</sup>lt;sup>2</sup> Categorisation is a cognitive mechanism which is not only explored in linguistics. An overview of some of the evidence for the importance of categorisation in cognition in general can be found in Harnad (2017).

frequency has on the mental grammar as *entrenchment*.<sup>3</sup> Moreover, in a similar way to Langacker (1987; cf. section 2.1.1), Bybee (2001: 35) rejects underlying representations in her conceptualisation of usage-based linguistics (Bybee 2001: 35). Rather, each individual token is stored in the speakers' mental grammar and categorised. In contrast to generative approaches, redundancies are not considered problematic (cf. the rule/list fallacy in section 2.1.3). In fact, redundant storage is one of the core assumptions of usage-based linguistics in general and CG in particular. This will be taken up in the following section again, which will consider the rule/list fallacy in more detail.

# 2.1.3. The rule/list fallacy

Usage-based approaches do not subscribe to the so-called rule/list separation (or, from a CG perspective, fallacy) typically assumed in generative linguistics (Bybee 2001: 7). The rule/list fallacy stems from the "assumption that rules and lists are mutually exclusive (Langacker 2000: 2). In other words, for reasons of economy, generative accounts do not allow lists of concrete expressions in the mental grammar if there exists a rule that can be used to derive those expressions. Thus, for instance, plural forms such as tables, glasses or books would not be represented in the grammar as lists, as there exists a general rule N + -s from which such items can be generated (Langacker 1987: 29). Usage-based approaches, however, reject this assumption and claim that generalisations are abstracted from stored, concrete expressions (Bybee 2001: 7). While including both schemas (i.e. generalisations) and instantiations (i.e. concrete expressions) (cf. redundant storage; section 2.1.2) in the grammar may not be especially economical, it generally provides a more psychologically accurate view of linguistic knowledge (Langacker 2000: 2); hence the maximalist, non-reductive and bottomup nature of CG. This is further substantiated by research in exemplar theory (e.g. Pierrehumbert 2001), which suggests that speakers of a language categorise and store every token they experience in large networks (Bybee 2006: 716). Moreover, it is assumed that from these stored tokens, generalisations of "various degrees of abstraction" (Bybee 2001: 7) are established. Put differently, the linguistic knowledge of a speaker consists of both concrete expressions as well as schematic units generalising over these expressions (Dabrowska 2004: 213). As was mentioned before, linguistic storage is considered "highly

<sup>&</sup>lt;sup>3</sup> As will be shown in later sections of this thesis, frequency effects prove pivotal in accounting for certain atypical patterns found in English vowel reduction.

redundant" (Langacker 2000: 2). Different schemas may describe the same structures at different levels of abstraction. Thus, CG acknowledges that "linguistic patterns occupy the entire spectrum ranging from the wholly idiosyncratic to the maximally general" (Langacker 1991: 263). The following section will look more closely at CG and lay out the theoretical foundations of the analysis that follows in subsequent sections.

#### 2.2. Grammar as a network of categories

The content requirement briefly discussed in the previous section has fundamental effects on the way grammar is conceptualised in CG. In the words of Langacker (1991: 263-264), grammar is defined as "a structured inventory of conventional linguistic units" (1991: 263-264). Since the rule/list separation is denied in CG, this structured inventory consists of both every concrete expression, i.e. instantiation, and generalisations, i.e. schemas, which have been abstracted from actually occurring expressions (Kumashiro & Kumashiro 2006: 80). Moreover, language, as emphasised by Langacker (2010: 108), is organised in complex, overlapping networks of different elements. A simple example of such a network-like structure is outlined in Figure 1 below:



Figure 1 Grammar as a network (based on Nesset (2008: 12))<sup>4</sup>

The boxes in Figure 1 symbolise cognitive schemas. Schemas are "the commonalities that [emerge] from distinct structures when one abstracts away from their points of difference [...]" (Langacker 2000: 4). A speaker may, for instance, experience many different utterances including different expressions for trees (cf. Nesset 2008). On the basis of his or her experience, a schema capturing the commonalities between these actually occurring utterances may then be abstracted, i.e. a schema stating that all of the instances in Figure 1

<sup>&</sup>lt;sup>4</sup> It may be reasonably objected that English does not have a nominative. However, it has been argued that English has a nominative-accusative alignment system (Keizer 2015: 195). While full noun phrases are not marked for either nominative or accusative, the distinction emerges with respect to pronouns (e.g. *He hit the man* vs. *The man hit him*). Consequently, the feature NOM is included in the the semantic poles of Figure 1.

refer to different types of trees. As a result, a network-like structure showing how the different schemas are related to each other emerges. The upper portion of the boxes in Figure 1 contain the semantic component of each unit, while the lower portion of the boxes specify the phonological form (2008: 11-12). The phonological form should not be understood as representing sound in the real world. Rather, as Langacker (1987: 79) puts it, it is an "auditory image" (1987: 79) and thus should be seen as a mental concept that summarises what many utterances of the word have in common (for discussion see Nesset 2008: 12). Schemas involving both a semantic and a formal component are Saussurean signs and called symbolic. It should again be noted at this point that CG does not subscribe to the rule/list separation mentioned in section 2.1. Thus, while schemas represent generalisations over actual occurrences, they are not independent from the units they are abstracted from (cf. Bybee 2001).

An important aspect of schemas that becomes evident in the discussion of Figure 1 is that they do not exist in an empty space in the mental grammar. Rather, schemas form networks and are connected to each other by categorising relationships. These are represented in Figure 1 above as arrows. In principle, there are two different types of such relationships, viz. instantiations and extensions (cf. Langacker 1987: 371; see also Langacker 2000: 4). The solid arrows are of the former type, the dashed arrows of the latter. Instantiations are characterised as a relation between two compatible schemas of which one shows a greater degree of specificity. Such a relation can be captured by the formula  $A \rightarrow B$ . It shows that the more specific schema B instantiates or elaborates the general schema A (Langacker 2000: 4). With respect to Figure 1, each of the bottom three schemas is more specific than that for 'tree'. It can easily be seen that all types of birches are also trees, but not all trees are birches. Hence, there are solid arrows from the top schema to each of the lower schemas. Moreover, note that the phonological pole of the top schema has to be empty, since "no salient phonological properties [...] recur in all the names of" trees (Nesset 2008: 13). The second type of relation, viz. extensions, describes a relation between two similar, yet to some degree conflicting schemas (Langacker 1987: 371; see also Langacker 2000: 4). The formula [A] --> (B) indicates that while (B) does not instantiate [A], it is nevertheless categorised by it. In Figure 1, there are two dashed arrows ranging from the BIRCH schema to the SYCAMORE and SHADBUSH schemas respectively. Birches can be considered relatively prototypical trees, but

not so much sycamores or shadbushes, which may be seen as rather peripheral. Thus, extensions always involve some kind of prototype against which other members of a category are compared (Langacker 2000: 4). As will be shown in later sections, both types of categorising relationships prove necessary in the proposed analysis of vowel reduction in SSB and Russian. The following section will address phonology in more detail and lays out the framework in which the subsequent analysis is couched.

#### 3. Cognitive Grammar and phonology

The aim of this section is to discuss some critical concepts in phonology and demonstrate how they can be explained in a CG framework. At this point it is necessary to return once more to the content requirement and its implications for the theory. One of the major advantages of cognitive linguistic approaches to language is their restrictiveness in terms of theoretical constructs (Nesset 2008: 13). Thus, only concepts not in disagreement with the content requirement (cf. section 2.1), e.g. cognitive schemas and categorising relationships, are allowed in CG. Moreover, these constructs are all cognitively motivated in so far as they are not restricted to linguistic phenomena, but instead are, as already emphasised, aspects of human cognition as a whole (Nesset 2008: 13; see also Bybee 2001). Consequently, the phonological theory outlined in this thesis requires a radically new way of thinking about phonology. Notions traditionally used in phonological analysis such as underlying representations or rules are prohibited by the theory through the content requirement (Nesset 2008: 13). Section 3.1 will briefly consider some basic notions in phonology, i.e. phonemes and phonological features. Section 3.2 will focus on issues crucial in the remainder of the thesis, viz. second-order schemas, schema interaction, well-formedness principles and the actualisation of candidate expression. The final subsection will outline some obvious parallels to another framework, viz. Optimality Theory.

## 3.1. Some basic phonological concepts in CG

#### **3.1.1.** Phonemes and allophones

Schema and network formation as outlined in the previous section is not limited to symbolic units. In order to perceive speech, it is necessary that "categories of acoustic events" (Taylor 2006: 21) are established which consider certain sounds to be identical in the phonological

system.<sup>5</sup> In accordance with its usage-based nature, CG assumes that exposure to usageevents allows language users to create phonological schemas capturing the commonalities of the sounds they hear (Nesset 2008: 31).<sup>6</sup> In this sense, phonemes are treated as conceptual categories that generalise over groups of phonetically similar but different sounds (Bybee 2001: 53). The act of categorising certain phonetic variants into one phoneme category hinges on phonetic similarity. Thus, two tokens may be classified as instantiating the same phoneme if they are sufficiently "similar in their acoustic (and articulatory) properties" (2001: 53). Figure 2 below illustrates such a network for the vowel /i:/ and three of its allophones. The diagram also includes four exemplary English words, i.e. conventional linguistic units, from which the schemas may be abstracted:



Figure 2 The phoneme as a complex category (based on Nesset's account of Russian vowels (2008: 32))

Figure 2 indicates how allophonic variation can be understood within a CG framework. Although this is a simplified representation, it can be seen that a nasalised [ĩ:] occurs immediately before nasalised consonants, while a slightly shortened [i·] appears before voiceless consonants (pre-fortis clipping; see Roach 2009: 28). The default allophone schema [i:] occurs elsewhere. Moreover, Figure 2 makes assumptions as to the status of the individual sounds in the phonemic system, viz. that [i:], [i·] and [ĩ:] are in complementary distribution.

<sup>&</sup>lt;sup>5</sup> The formation of phonological categories is supported by research in cognitive sciences. See, for example, Goudbeek et al. (2017). For a (linguistic) discussion of the concept of the phoneme as a conceptual category, see Mompeán-Gonzáles (2004).

<sup>&</sup>lt;sup>6</sup> While phonological schemas do not necessarily contain a semantic pole, they are not at variance with the content requirement. Phonological structures are one of the three types of structures allowed in CG (cf. section 2.1).

They never occur in the same context and therefore cannot change meanings in minimal pairs. Thus, they can be considered allophones (cf. Nesset 2008: 32-33).

Figure 2 also exemplifies the bottom-up approach of cognitive grammar mentioned in section 2.1.1. The concrete instantiations experienced by language users on a daily basis give rise to allophone schemas. However, speakers may also generalise over the allophone schemas including the high front vowel [i:] and form a more abstract schema, viz. the default allophone schema (Nesset 2008: 32). As can be seen, [i:] appears in all contexts except before nasalised and voiceless consonants, which is indicated in Figure 2 above by the suspension points. Consequently, it can be regarded as the elsewhere case. Moreover, speakers may generalise over all three allophones and form a schema on a higher degree of abstraction, i.e. the phoneme schema, which captures the fact that [i:], [i'] and  $[\tilde{i}:]$  are all high vowels. This is in line with the assumption discussed in section 2.1 that "linguistic knowledge is represented at varying degrees of abstraction" (Dabrowska 2004: 213) and storage is highly redundant (cf. Bybee 2001; Langacker 2000). As Langacker (1987: 389) puts it, "the emergence of a phoneme [...] is [...] a process of decontextualization" (Langacker 1987: 389). High vowels can occur in all environments and form minimal pairs with low and mid vowels. It follows that phoneme schemas have to be specified in a context-free manner (as opposed to allophone schemas, which are generally specified in terms of context). Nesset (2008: 33) stresses that the view of phonemes as "categories of related sounds" (2008: 33) is compatible with traditional notions of the phoneme as a psychological unit (cf. Anderson 1985). Figure 2 makes use of the phonological schema [high]. Since features may be considered problematic in usage-based approaches, they will be discussed in the following section in more detail.

# 3.1.2. Phonological features in CG

The use of phonological features in CG may seem controversial at first, as they have been traditionally employed in generative approaches to phonology (e.g. Chomsky & Halle 1991 [1968]). In such frameworks, phonological features are taken to be part of Universal Grammar: "[T]hey represent the phonetic capabilities of man and, we would assume, are therefore the same for all languages" (1991 [1968]: 295). This innate and universal set of features is used to describe all patterns and structures found in phonology, such as, for example, natural classes (Cohn 2011: 19-20). Moreover, generative approaches characterise each feature by its binary nature. Put differently, to define phonological inventories of various

languages, features can either be positively or negatively specified (Cohn 2011: 20; cf. Chomsky & Halle 1991 [1968]). However, this conception of features is inherently at variance with the content requirement stated in section 2.1 above, which only allows phonological structures that occur in language use. In other words, features as conceptualised by generative phonology are precluded by the theory and consequently cannot be part of a CG analysis of phonological phenomena. This naturally raises the question as to how natural classes or other groups of sounds that behave alike are dealt with in CG. Regarding phonological features as generalisations speakers make over usage-events resolves this issue.

In line with more recent frameworks emphasising the emergent nature of language (e.g. Blevins 2004; Mielke 2008 and others), CG views phonological features to emerge from actual usage (cf. Langacker 2000; Nathan 1996; Nesset 2008). Thus, features are conceptualised as "abstract categories based on generalizations that emerge from phonological patterns (Mielke 2008: 9) or, as Langacker (2000: 44) puts it, as "schematic characterisations of 'natural classes' of sounds" (2000: 44). As can be seen in Figure 2 in section 3.1 above, it may be said that speakers form a schema [high] to generalise over the three high vowels [i:], [i'] and [ĩ:] (of course, other vowels such as /u:/ may be subsumed under the same general schema [high]). Furthermore, as phonological features are generalisations grounded in language use, it follows that they must be positively specified. Put differently, schemas do only exist for structures that can be found in language use. Since speakers do not experience [-high] vowels, such a schema can logically not be established (cf. Nesset 2008: 36).<sup>7</sup> Moreover, treating features as schemas allows for redundancy, which, as already mentioned, is one of the core assumptions of CG (cf. Bybee 2001; Dabrowska 2004). Furthermore, it should be pointed out that the theory does not rule out the possibility that a particular sound is categorised by more than one schema. The three allophones in Figure 2 may not only be subsumed under the schema [high], but presumably also under [front] or [unrounded].<sup>8</sup> What becomes apparent in this discussion is that phonological

<sup>&</sup>lt;sup>7</sup> Underspecification theory offers a possible alternative to negatively specified features. As pointed out by Nesset (2008: 36), treating certain vowels as not specified for a particular feature, e.g. frontness, is not at odds with the content requirement. For instance, instead of using the feature [-front], it may be said that the English short vowels / $\sigma$  p  $\Lambda$ / are not specified for frontness at all. This, however, is not accepted by all scholars in the field. Nathan (1996: 109), for instance, argues against underspecification theory in CG by claiming it is fundamentally at odds with its principles.

<sup>&</sup>lt;sup>8</sup> The feature [unrounded] should not be understood as a negatively specified feature. Rather, it stands for a spread lip position (cf. Nesset 2008: 36).

categories or schemas form similar networks such as those outlined in section 2.2 for the mental grammar as a whole. This once more underlines the non-modular approach CG takes (cf. section 2.1). Phonology is not taken care of by a particular phonological component of the grammar, but rather is part of wider linguistic and non-linguistic cognition. Having outlined fundamental phonological issues in CG terms, the thesis will now move to the discussion of core concepts in CG needed in the analysis of English vowel reduction.

# **3.2.** Key concepts in CG phonology

#### 3.2.1. First- and second-order schemas

First- and second-order schemas are crucial in a CG analysis of phonological phenomena, as they allow for generalisations covering (morpho)phonological alternations (cf. Nesset 2008: 20). Consider Figure 3 below, which shows two first-order schemas (left) and a second-order schema (right) for a hypothetical language in which vowels preceded by a palatalised consonant reduce to [1] in unstressed syllables:



Figure 3 First and second-order schemas exemplified

First-order schemas can most adequately be characterised as generalisations over actually occurring utterances. For instance, as shown in the two left boxes in Figure 3, speakers may form first-order schemas over vowels in stressed and unstressed syllables. Nesset (2006: 56) further stresses that "language users may compare vowels in stressed and unstressed syllables and connect them by means of categorizing relationships" (2006: 56). Since the two first-order schemas in Figure 3 are partly compatible with each other, they can be connected using an extension relation symbolised by the dashed arrow (cf. section 2.2). If such relations between stressed and unstressed vowels recur in systematic ways, speakers may form larger schemas, i.e. second-order schemas, capturing these relationships (2006: 56). In other words, second-order schemas can be understood as "schemas over schemas that are connected via categorizing relationships" (Nesset 2008: 19).

The two terms product-oriented and source-oriented generalisations further illustrate the difference between first- and second-order schemas. First-order schemas, i.e. the two schemas to the left in Figure 3, capture what is commonly referred to as product-oriented generalisations (Nesset 2008: 20). Product-oriented schemas do not relate the structures to any underlying source. Rather, as Bybee (2001: 126) puts it, they "[generalize] over forms of a specific category, but [do] not specify how to derive that category from some other" (2001: 126). Thus, the first-order schemas in Figure 3 only generalise over the quality of a vowel in stressed and unstressed syllables, but do not state that reduced vowels are derived from full vowels. By contrast, source-oriented generalisations focus on the source to which particular rules are applied to create certain well-formed surface structures (Nesset 2008: 20). Accordingly, source-oriented generalisations (e.g. rules) enjoyed great popularity in rulebased frameworks in which underlying representations play an important role (e.g. Chomsky & Halle (1991 [1968]). In CG, it is second-order schemas which enable source-oriented generalisations to be captured without employing underlying representations. Thus, the second-order schema to the right in Figure 3, for instance, takes the fully realised vowel in stressed syllables as the source, which is then related to the reduced vowel in unstressed syllables. Put differently, two actually-occurring structures are related to each other by means of categorising relationships. It should further be noted that product-oriented schemas precede source-oriented generalisations (Bybee & Slobin 1982: 288). In order to establish relations between stressed and unstressed syllables (and hence source-oriented schemas), a speaker must have formed generalisations over stressed and unstressed syllables in isolation first. Having established the schema types needed in the analysis, the thesis will now examine schema interaction in more detail.

# 3.2.2. Schema interaction and well-formedness principles

The different ways in which schemas can interact with each other are crucial in CG. The version of CG (cf. Kumashiro 2000; Nesset 2008) used in the present thesis assumes that speakers can "employ [alternative strategies] when activating schemas in categorization networks" (Nesset 2008: 14).<sup>9</sup> The analysis to be presented in later sections of this thesis

<sup>&</sup>lt;sup>9</sup> This refers to the assumption that speakers may actualise a variety of candidate expressions (based on the cognitive schemas they extract from language use) from which one is chosen as the winning candidate. Thus, the theory presented here models the sanctioning of possible alternatives and how cognitive principles may solve the resulting competition.

models these strategies. However, since the phenomena examined are relatively complex, the theoretical aspects concerning schema interaction are illustrated using a simple, hypothetical example. Consider a language in which all vowels reduce to [I] after palatalised consonants, while in all other contexts, they reduce to [ə]. With respect to this rather simple vowel reduction pattern, two possible schemas can be established. These are shown in Figure 4 below:



Figure 4 Schema interaction exemplified

Figure 4 presents an illustration of schema interaction in CG. In this hypothetical example, two different schemas are contained in the mental grammar. The left second-order schema captures reduction to [1], the right second-order schema reduction to [2]. A speaker may wonder about the pronunciation of the hypothetical word t/ap once a stress-shifting suffix (- $\delta s$  in the present example) is added to the stem. Since there are two schemas in the mental grammar, two competing candidate expressions, i.e. alternatives, can reasonably be given. They are illustrated in round boxes in Figure 4. Candidate expressions in CG are "hypotheses that speakers and hearers can make about their native language" (Nesset 2008: 14).

To resolve the competition between candidate expressions, language users compare the alternatives to the schemas in their mental grammar (Nesset 2008: 15). Each candidate expression in Figure 4 instantiates one schema. The left candidate is categorised by the schema that states that vowels after palatalised consonants reduce to [1] when stress is lost. By contrast, the right candidate expression instantiates the reduction schema generalising over reduction to [ə]. It should be emphasised, however, that, the candidates differ with respect to the degree to which they fit their respective schemas. In other words, the left candidate is categorised by the schema which states the context of reduction, while the candidate to the right instantiates the less specific, more general schema. The competition in simple examples such as these is decided on the basis of the notion of *conceptual overlap* (Langacker 1999: 106). Langacker (1999: 106) explains the concept as follows:

[One] factor [in how such competitions can be resolved] is the amount of overlap between the target and a potential categorizing structure. We can reasonably assume that the sharing of features is what enables the target to stimulate members of the activation set in the first place, and that the degree of stimulation is roughly proportional to the number of features shared. [L]ower-level Schemas, i.e. structures with greater specificity, have a built-in advantage in the competition with respect to higher-level Schemas. Other things being equal, the finer-grained detail of a low-level schema affords it a larger number of features potentially shared by the target.<sup>10</sup>

As can be seen in Figure 4 above, the left candidate complies with the more specific schema. Consequently, it conceptually overlaps with its schema to a higher degree than the other candidate (i.e. it instantiates the more specific schema) and is chosen as the winner.<sup>11</sup>

Although conceptual overlap is a pivotal concept in schema interaction, it does not suffice in more complex reduction patterns. Consequently, other criteria need to be added to account for the phenomena analysed in this thesis. Based on a working hypothesis suggested by Langacker (1991), Kumashiro (2000: 25) proposes four principles, which govern the well-formedness of candidate expressions. They are given in (1) below:

(1) Well-formedness principles (Kumashiro 2000: 25)

<sup>&</sup>lt;sup>10</sup> Langacker (1999: 105) also discusses two additional factors, viz. entrenchment and contextual priming. Entrenchment will be taken up again when discussing exceptions to the patterns found in the data set. Contextual priming, i.e. cues in the discourse context, is not of relevance to the issues discussed in the thesis.

<sup>&</sup>lt;sup>11</sup> It is interesting to point out that such situations are well-known in linguistics. Within generative frameworks, such situations have typically been dealt with by the *Elsewhere Condition* (cf. Kiparsky 1982), which guarantees that more specific rules prevail over general rules.

a. Access

When a given candidate expression is assessed relative to a certain subpart of the grammar, i.e. a function, categorizing units (from the network representing the subpart) that are schematic to, or are elaborated by, the expression are activated and sanction the expression.

b. Activation

The total 'activation', i.e. conventional motivation/sanction, of a candidate expression is the sum of the activation values obtained from all of the categorizing units. Each such value correlates positively with the expression's 'distance' from the unit, i.e. how far it diverges from its categorizing unit by elaboration.

c. Uniqueness

When there are multiple candidate expressions, all but the one with the highest activation value are deactivated.

d. Well-formedness

The degree of well-formedness of a candidate expression correlates with its final activation value.

These well-formedness principles determine the winning candidate expression in cases in which conceptual overlap alone fails to do so. Principle (1a) assures the actualisation of candidate expressions outside the grammar. It follows that the two candidates in Figure 4 are by no means arbitrarily postulated. Rather, they emerge from core principles in CG and are sanctioned by categorising units in the grammar. The second principle (1b) concerns the activation, i.e. conventional motivation, of candidate expressions. The total activation of a candidate refers to its likelihood of winning the competition. A high total activation greatly enhances the chances of an expression to be selected as the winner.<sup>12</sup> A candidate expression obtains activation value from each categorising unit it is categorised by. In other words, the more schemas an expression instantiates, the higher its activation value. Moreover, Langacker's notion of conceptual overlap (1999: 106) is implied in (1b) as well. A candidate that exhibits a higher degree of overlap with a unit (i.e. a low-level schema) can be assumed to be 'closer' to that categorising unit than to a highly abstract schema. Thus, additional activation value is added to a candidate that way. The third principle in (1c) guarantees that

<sup>&</sup>lt;sup>12</sup> The notion of activation value is relative. What constitutes high activation value always depends on the activation values of other candidates in the competition.

only one of the actualised candidates, viz. the expression with the highest activation value, is activated, while all others are deactivated.<sup>13</sup> The last principle in (1d) is a rather general statement on the gradient nature of well-formedness in linguistics and its positive correlation with the total activation value.<sup>14</sup> At this point, certain similarities to Optimality Theory (OT) become apparent. While section 3.2.3 is not meant to discuss these in any detailed way, a short overview of OT's basic principles will prove useful.

# **3.2.3.** Some parallels to Optimality Theory

In the previous subsection, a number of parallels between the model of CG used in this thesis and Optimality Theory (OT) have become evident. In both theories, candidate expressions are compared to the mental grammar a speaker holds. While in OT the candidates are evaluated with respect to well-formedness constraints (cf. Kager 1999; Prince & Smolensky 2004), CG captures constraints in cognitive schemas. Moreover, in both approaches, candidate expressions are compared to the entire grammar simultaneously (Nesset 2008: 17). Despite points of similarities, however, a number of fundamental differences can be identified. In OT, there are two types of constraints, viz. markedness and faithfulness constraints (Kager 1999: 9). Markedness constraints are needed in the framework to assure that the output is well-formed in terms of structure. Faithfulness constraints regulate that input and output forms remain to some degree similar (1999: 9). All constraints are universal in that they are part of Universal Grammar, but, at the same time, they are ranked in language-specific ways (1999: 11). Such constraints do not exist in CG. In CG, schemas are considered to be generalisations that speakers form on the basis of exposure to language. Consequently, they have to be language-specific and cannot be universal.

Further differences can be found in connection with CG's commitment to the content requirement. In OT, faithfulness constraints require some underlying representation against which the surface form is evaluated (Kager 1999: 9). However, as CG is a usage-based approach, it does not recognise any structures not attested in actual speech. As discussed in section 3.2.1, source-oriented generalisations are captured in CG by second-order schemas relating two actually-occurring structures to each other. Furthermore, constraints in OT are

<sup>&</sup>lt;sup>13</sup> (1c) should by no means be considered an absolute principle (Kumashiro 2000: 25). In some cases, two alternative ways of pronunciation exist (e.g. *garage* /'garidʒ/ vs. /'gara:(d)ʒ/).

<sup>&</sup>lt;sup>14</sup> Kumashiro argues that these principles are essentially compatible with connectionist approaches to language. For an overview of connectionist models, see Pulvermüller (2001).

often negatively stated (e.g. Vowels must not be nasal; Kager 1999: 1). In CG, however, schemas must be specified positively, since they cannot be abstracted from something that does not occur in actual utterances (cf. section 3.2.1). Differences can also be found in the way the winning candidate expression is selected. As already mentioned, universal constraints in OT are hierarchically ranked in a language-specific manner. The phonological framework outlined in the previous sections, however, selects the winning candidates based on rather general cognitive principles such as categorisation relationships and schema interaction. Having discussed the the theoretical underpinnings of the subsequent analysis, the thesis will now turn to the variety studied.

#### 4. The sound system of Standard Southern British

This section will discuss the variety of British English analysed in the present thesis, viz. Standard Southern British (SSB). Section 4.1 will define SSB in contrast to the traditional standard Received Pronunciation (RP). In section 4.2 the sound system will be examined in more detail and some of the major contrasts to RP will be outlined.

## 4.1. RP and Standard Southern British

Received Pronunciation has a long history in Great Britain. Starting in the 19<sup>th</sup> century in the area of London, RP's rise continued until the 20<sup>th</sup> century (cf. Lindsey 2019). It is defined as the variety traditionally spoken by the upper and upper-middle classes of British society or, more precisely, "as the educated pronunciation of the metropolis, of the court, the pulpit, and the bar" (Ellis 1869: 23). The invention of mass communication in the first half of the 20<sup>th</sup> century advanced the spread of RP throughout the United Kingdom (2019: 3). However, from the 1960s onwards, RP started losing much of its formerly prestigious status. This development, as Lindsey (2019: 3) points out, was closely connected to the democratisation of British society. Moreover, already in the 1980s, Wells (1982, 1: 118) speculated that "[w]ith the loosening of social stratification and the recent trend for people of working-class or lower middle-class origins to set the fashion in many areas of life, it may be that RP is on the way out" (1982, 1: 118). While this prognosis has proved to be true in the long run, the south of England and especially London remain the centre of power and wealth (Lindsey 2019: 4). A variety of different accents can be heard in the media in present-day Britain, but accents of the South, especially spoken by the middle and upper-middle classes, are still

dominant. It is accents of this sort that are typically referred to as British English or Standard Southern British (SSB) (2019: 4).

Standard Southern British is a variety "characteristic of university-educated young adults from the south of England" (Lindsey & Szigetvári 2013).<sup>15</sup> While it cannot be considered a majority accent in the UK, SSB certainly is one of the accents English speakers all around the globe are most familiar with. It is frequently used by presenters on British TV channels and can also be heard in Hollywood films. Thus, in the minds of many, SSB is closely linked to Britain internationally (Lindsey & Szigetvári 2013).<sup>16</sup> Some general features of SSB include a tendency towards converging spelling and pronunciation and towards making the pronunciation of many words more similar to American English (Lindsey 2019: 9-12). The former can again be understood with respect to the democratisation of British society. For instance, interest was pronounced as /intrist/ in RP, but nowadays the DRESS vowel  $\epsilon$  in the second syllable can be found in the pronunciation of most speakers (2019: 10). The latter is a direct consequence of the cultural influence of the US. Among such changes there are unstressed yod coalescence (cf. section 4.2.2), weak vowel merger or vocal fry. It should be noted though that SSB retains many of its distinct British characteristics such as, for example, non-rhoticity or the LOT vowel (2019: 11). The sound system of SSB will examined in more detail in the following section.

#### 4.2. The sound system of SSB

#### 4.2.1. Vowels

Many of the changes that occurred in SSB can be found in its vowels. In fact, Lindsey (2019: 18) argues for "a large-scale 'anti-clockwise' shift in the vowel system" (2019: 18), which moved the RP vowels to different positions in the vowel space. While the number of contrasts largely remained identical, the observed shifts mostly concern vowel quality. They are illustrated in Figure 5 below (based on Lindsey 2019: 18):

<sup>&</sup>lt;sup>15</sup> See http://cube.elte.hu/accent.html (1 Dec. 2019).

<sup>&</sup>lt;sup>16</sup> See http://cube.elte.hu/accent.html (1 Dec. 2019).



Figure 5 Anti-clockwise vowel shift in SSB (based on Lindsey 2019: 18)

Figure 5 shows the changes in vowel quality that can be observed in SSB with respect to their traditional RP pronunciation. As can be seen, all front vowels were lowered as a consequence of the vowel shift. This can most clearly be observed in the lexical sets DRESS and TRAP, which show the more open [ $\epsilon$ ] and [a] rather than the RP vowels [e] and [æ]. Lindsey (2019: 19) further notes that the starting points of the diphthongs in FLEECE<sup>17</sup> and FACE have moved to a lower position in the vowel space closer to [I] and [ $\epsilon$ ]. Moreover, the diphthong of PRICE has a starting point further back in SSB, which can be transcribed as [a].

Back vowels have experienced two different types of changes, viz. raising and centralisation (cf. Lindsey 2019: 20-21). The LOT and THOUGHT vowel have been raised and are pronounced with a vowel closer to [5] and [6] respectively in SSB. It should also be mentioned that the vowel in THOUGHT, NORTH, and FORCE are identical in southern Britain. Thus, *saw* and *sore*, as well as *caught* and *court* are pronounced in the same manner, i.e. [so:] and [ko:t] (2019: 20). With respect to centralisation, Figure 5 shows that the lexical sets FOOT and CURE are nowadays pronounced with a more central vowel [ $\theta$ ]. The starting points of the diphthongs in GOAT and GOOSE have also been centralised (2019: 20).<sup>18</sup> What should be noted at this point is that, as a result of vowel shifts, the vowel system as presented in this

<sup>&</sup>lt;sup>17</sup> The vowel in FLEECE has traditionally been described as a close monophthong /i:/ (Lindsey 2019: 23). Gimson (1974 [1970]: 99), however, already indicates a noticeable diphthongisation of /i:/ in RP. In SSB, it is typically pronounced as a diphthong [ii] or [ij] (Lindsey & Szigetvári 2013; see http://cube.elte.hu/accent.html (1 Dec. 2019)).

<sup>&</sup>lt;sup>18</sup> Similar to FLEECE, GOOSE has traditionally been treated as a close monophthong /u:/ (Lindsey 2019: 23). Gimson. (1974 [1970]: 120) indicates diphthongisation for /u:/ as well. In SSB, it is pronounced as a diphthong [uu] or [uw] (Lindsey & Szigetvári 2013; see http://cube.elte.hu/accent.html (1 Dec. 2019)).

They can be grouped into two categories, i.e. those ending in the glide [j] and those ending in the approximant [w] (Lindsay 2019: 24): **Table 1** Diphthongs in SSB

Diphthongs	ending in [j]	Diphthongs ending in [w]		
FLEECE	/ıj/	GOOSE	/uw/	
FACE	/ɛj/	GOAT	/əw/	
PRICE	/aj/	MOUTH	/aw/	
CHOICE	/oj/			

The vowels in Table 1 form a natural class. They behave alike in various phonological processes, such as smoothing or pre-fortis clipping. Moreover, as opposed to monophthongs, they can directly precede other vowels (e.g. *chaos* /kéjɔs/ or *flower* /fláwə/) (2019: 24). A systematic chart of the vowel system of SSB, juxtaposed with its RP equivalents, will be given in section 5.1 when discussing transcription practices for SSB. The following section will consider some aspects of the consonant system of SSB. However, since consonants are not crucial to the phenomena analysed in this thesis, only an overview is given.

### 4.2.2. Consonants

Most differences in the sound systems of SSB and RP can be found in the vowels. However, a brief examination of some of the major changes in the consonant system will prove useful as well. One of the tendencies found in SSB is strong aspiration of /p t k/ in stressed and unstressed syllables (Lindsey 2019: 55). This stands in contrast to RP, where aspiration of these plosives traditionally only occurred in stressed syllables. Another difference worth mentioning is the palatalisation (or yod-coalescence) of RP's consonant clusters /tj/ and /dj/ resulting in SSB in /tf/ and /dʒ/ respectively. Although more frequently found in weak syllables (e.g. *education*), Lindsey (2019: 59) argues that it can increasingly be heard in stressed syllables as well (e.g. *Tuesday*). It should be mentioned that palatalisation can also be found in some words in RP (e.g. *culture* and *future*). However, as Wells (1982, 1: 247) puts it, the sound /tf/ in words like *situation* in RP sounds "rather vulgar" (1982, 1: 247). Nevertheless, coalescence is widespread in SSB nowadays and not negatively connotated anymore (Lindsey 2019: 59). A final note on consonants in SSB concerns the use of the syllabic consonants /l/ and /n/. These are less frequent in the speech of speakers of SSB than

in traditional RP. While words such as *bottle* or *curtain* were pronounced [botl] and [kə:tn] in RP, a weak vowel frequently follows the plosive in SSB, e.g. [botəl] and [kə:tən] (2019: 65). Having presented the sound system of SSB, the thesis will now move on to a number of methodological aspects to be considered.

# 5. Methodological aspects

The aim of this section is to discuss all the aspects of the study related to data and data collection. Since the dictionary that was used in the present study uses of a modified IPA system, section 5.1 will briefly discuss the transcription conventions applied in the remainder of the thesis. In section 4.2, the data collection process will be outlined in detail.

#### 5.1. Dictionary and transcription symbols

The dictionary used to gather the data analysed in the subsequent study is called *CUBE: Current British English searchable transcriptions* (Lindsey & Szigetvári 2013). As discussed in section 4, the pronunciation of SSB has changed in many aspects in comparison to traditional standard RP. However, although most commonly found in the dictionaries on the market (e.g. Wells 2008), the traditional transcription system introduced by Gimson (1974 [1970]) for RP does not accurately represent 21st century SSB and its changes (Lindsey & Szigetvári 2013).<sup>19</sup> This is particularly evident in the choice of symbols for vowels, which have arguably undergone the most radical changes, viz. the anti-clockwise vowel shift (cf. section 4.2.1). As a result, the accent described by the traditional vowel symbols "now sounds to native speakers old-fashioned, 'posh' and even amusing" (Lindsey & Szigetvári 2013).<sup>20</sup> To reflect contemporary changes in the pronunciation of British English, the transcriptions used by CUBE differ from classical RP in several ways. These are shown in Table 2 below. Since the consonant system essentially remains the same, only the vowel symbols and their RP equivalents for comparison (given in grey) are shown (based on Lindsey 2019: 146):

<sup>&</sup>lt;sup>19</sup> See http://cube.elte.hu/accent.html (1. Dec. 2019)

<sup>&</sup>lt;sup>20</sup> See http://cube.elte.hu/accent.html (1. Dec. 2019)

short vowels		long vowels									
			long monophthongs				diphthongs				
KIT	I		NEAR	Ľ	IÐ	FLEECE	ıj	i			
DRESS	8	е	SQUARE	<b>£</b> ]	eə	FACE	εj	еі			
LOT	э	υ	THOUGHT	0:	<b>0</b> ]	CHOICE	oj	JI			
TRAP	a	æ	PALM	aː		PRICE	aj	аі	MOUTH	aw	aʊ
FOOT	θ	Ω	(CURE	θΪ	ບອ)				GOOSE	<del>u</del> w	u
STRUT	Λ		NURSE	ə:	31				GOAT	əw	90 90
comma ə											

Table 2 Transcription system for vowels in SSB (based on Lindsey 2019: 146)

Table 2 briefly summarises the major shifts in the vowel system already discussed in section 4.2.1. As can be seen, the symbols of all but three vowels differ from their RP equivalents. The centring diphthongs (NEAR, SQUARE, CURE) in RP are pronounced as long monophthongs in SBB (Lindsey & Szigetvári 2013).<sup>21</sup> It should be noted though that the vowel in CURE is frequently pronounced with the vowel used in THOUGHT (e.g. *pour* /pó:/), hence the brackets in Table 2. The following section considers the type of data collected.

# 5.2. Type of data

Words were collected on the basis of two criteria, viz. word length and derivational morphology. In other words, the present thesis examines word pairs that consist of a base word in which a given syllable is stressed and a derivative in which stress moves to a different syllable as a result of affixation (e.g. able - ability). While the number of syllables for base words was not significant, the length of the derivatives was a deciding factor, viz. derivatives had to have at least two syllables. Collecting word pairs allowed for not having to stipulate underlying representations, which are not compatible with the principles of usage-based approaches and the content requirement. Rather, by comparing base words and their respective derivatives, vowel reduction processes could be directly observed without resorting to underlying representations. The affixes selected to collect the derivatives are based on Giegerich's (1999) distinction between stress-neutral and stress-shifting affixes.

<sup>&</sup>lt;sup>21</sup> See http://cube.elte.hu/accent.html (1. Dec. 2019)

Apart from stress-shifting behaviour, no selection criteria were employed in choosing the affixes.

During the data collection process, some problems concerning the transcription system used by CUBE arose. A first glance at the original data did not show any systematic distribution of the reduced vowels [ə] and [1] for the vowels  $/\epsilon$ / and /qj/. At a later stage of the research, the transcriptions were checked once again. At this point, CUBE had changed the [ə] in most of the words to [1]. These changes were incorporated in the data set. However, other issues such as differences in the unstressed vowel between the singular and plural of the same word, e.g. [kómpjʉwtɛjʃən] vs. [kómpjətɛjʃənz], still remain. Where necessary, these issues will be discussed in the sections examining the data in more detail. As of now, it suffices to note that, while carefully compiled, the data set may still contain items that strike the reader as inconsistently transcribed. CUBE is continually updating and correcting its transcriptions. Nevertheless, changes incorporated into the dictionary only recently may not be reflected in the data set considered in the analysis. Having discussed some of the methodological issues, the thesis will now move on to a discussion of vowel reduction in SSB and introduce the data set in more detail.

#### 6. Vowel reduction in SSB

Before the data can be adequately discussed, stress and vowel reduction in general will have to be examined in section 6.1. Section 6.2 will present the data set and discuss exceptions to the patterns identified.

#### 6.1. Stress and vowel reduction

Since vowel reduction only occurs in unstressed syllables, a brief look at stress proves necessary. The languages of the world broadly fall into two groups with respect to stress placement, viz. fixed-stress languages and variable-stress languages (van der Hulst 2010: 33; see also Beckman 1986). In fixed-stress languages, primary stress is always placed on the same syllable in the word (e.g. on the first syllable in Slovak). By contrast, the position of the stressed syllable in variable-stress languages is not fixed and depends on a variety of other factors (2010: 33). Contemporary English belongs to the latter of the two groups and has further been termed a stress-accented language (Tokar 2017: 37). In stress-accented languages, the position of the accented syllable is typically determined on the basis of a

variety of cues such as vowel quality, intensity and duration (2017: 37).<sup>22</sup> Thus, according to Gussenhoven (2006: 216), "the most prominent syllable in [an English] word is longer than other syllables and has a less reduced vowel [...]" (2006: 216). An additional factor in the placement of stress in languages such as English or Modern High German is syllable weight (Tokar 2017: 38). While syllables are traditionally divided into heavy and light syllables, what counts as heavy or light (and therefore stressable) is often language-specific (2017: 38). For English, CV can be considered light, while all other syllable templates, i.e. CVC, CVV, CVVC and CVCC, are heavy, i.e. they show a branching rhyme (Wenszky 2004: 11-12). Quite generally, stress in English may in principle either fall on the penultimate or antepenultimate syllable in nouns and on the final or penultimate syllable in verbs (cf. Chomsky & Halle 1991 [1968]).

The quality of vowels is typically connected to stress placement. Many languages do not sustain the full vowel inventory in all possible contexts (Harris & Lindsey 2000: 190). Rather, stress-timed languages, e.g. English, tend to neutralise vowel contrasts in unstressed positions (cf. Crosswhite 2001). In stress-timed languages, the time intervals between stressed syllables tend to be equally long (Fox 2000: 88). What follows is that this necessarily results in the reduction of unstressed syllables to keep the length of the time intervals roughly the same. Consequently, unstressed syllables are typically shorter than their stressed counterparts, have a lower amplitude and do not show any notable pitch contour (Walker 2011: 16). Furthermore, with respect to positional neutralisation, two patterns can be identified, viz. prominence-reducing and contrast-enhancing vowel reduction (2001: 21). Prominence-reducing, i.e. centripetal, vowel reduction draws vowels into a more central position in the vowel space, e.g. vowel reduction in Romance languages (Harris & Lindsey 2000: 190). Conversely, contrast-enhancing, i.e. centrifugal, vowel reduction reduces the contrasts found in unstressed positions by dispersing vowels to the corners of the vowel space, e.g. vowel reduction in Russian. The reduction processes in SSB to be presented in section 6.2 and analysed in section 7 mostly conform to the centripetal reduction pattern, i.e. vowels reduce to a schwa-like quality (cf. Crosswhite 2001: 205 for English in general). However, Harris & Lindsey (2000: 190) note that languages characterised by centripetal reduction frequently also show aspects of centrifugal patterns (see also Harris 2005). As will

<sup>&</sup>lt;sup>22</sup> This stands in opposition to pitch-accent languages like Japanese, in which pitch is taken as the primary cue for accent (Tokar 2017: 37; see also van der Hulst 2010: 11-12).

be shown in the following section, SSB not only exhibits reduction to [ə] (cf. Crosswhite 2001 on English), but in some cases also to [1], viz. it also shows centrifugal patterns.

### 6.2. Data and discussion

This section will introduce the data considered in the analysis. However, since the data set is relatively large, only a subset will be shown at this point. The entire data set collected for the present thesis can be found in section 11 (appendix A). In order to make the presentation of the data easier, each subsection will focus on one type of vowel found in SSB, viz. short vowels (section 6.2.1), long monophthongs (section 6.2.2) and diphthongs (6.2.3). The final subsection will give a brief overview of the different patterns identified.

#### 6.2.1. Short vowels

#### 6.2.1.1. The data

The examples in Table 3 illustrate the reduction of each of the six short vowels in SSB. First, two examples each are given in which reduction occurs, followed by instances in which the full quality of the vowel is retained.

$\mathfrak{I} \rightarrow \mathfrak{I}$	3	$\mathfrak{e}  ightarrow 3$	$I \leftarrow 3$	8
pózıt	tóksık	əpélət	édıt	féstiv
pəzí∫ən	təksísətıj	ápəlέj∫ən	ıdí∫ən	fɛstívətıj
əbʻəlı∫	hóstajl	segmént	trépid	ıkspékt
ábəlí∫ən	həstílətıj	ségməntéj∫ən	trıpídıtıj	ékspɛktéj∫ən
$V \rightarrow \mathfrak{I}$	Λ	$a \rightarrow \vartheta$	a	
káridz	abdákt	rápid	káptīv	
kəréjdzəs	ábdʌktíj	rəpídətıj	kaptívətıj	
kənsált	páblik	ád	áktīv	
kónsəltéj∫ən	рлblísətıj	ədí∫ən	aktívətıj	
I		θ		
áktıv		føl		
aktívətıj		fəlfilmənt		

 Table 3 Reduction of short vowels

Table 3 shows that full vowels may occur in both stressed and unstressed syllables. Moreover, it can be seen that vowel reduction in SSB is a relatively strong centripetal, prominence-reducing process, as a result of which the vowels are drawn to a more central position in the vowel space. Note, however, that  $/\epsilon$ / exhibits two different outcomes, viz. [ə] and [I]. As already mentioned in the previous section, languages following a centripetal pattern often show centrifugal, or contrast-enhancing, reduction as well. What can further be seen in Table 3 is that /I/ and  $/\Theta$ / do not reduce. While no data was found to substantiate the claim for  $/\Theta$ / (i.e. only one word were found for  $/\Theta$ /), it is nevertheless assumed on the basis of Crosswhite's survey of vowel reduction in English (2001: 205). In her analysis, she argues that English exhibits two variants in reduction: either all unstressed vowels or all but /I/ and  $/\Theta$ / reduce to [ə] (2001: 205). This view is further supported by cross-linguistic tendencies for /I/ and  $/\Theta$ / not to reduce, such as can be found in Russian (cf. section 8), Brazilian Portuguese (Pöchtrager 2018) or Eastern Catalan (Harris 2005). The data set clearly shows that /I/ does not reduce in SSB.<sup>23</sup> Consequently, the present thesis assumes that  $/\Theta$ / does not reduce either. The data in Table 3 further show that full vowels can also occur in unstressed syllables. The factors that condition the retention of the full vowel quality in unstressed position will be discussed in the following section.

# 6.2.1.2. Open vs. closed syllables

The data presented in section 6.2.1.1 suggest that lack of stress is not the only factor governing the reduction of vowels in SSB. A clear distinction exists between open and closed syllables. Several authors (Burzio 1994: 112-126; Fudge 1984: 200; Marchand 1969: 222-225) conclude that unstressed short vowels in open syllables generally reduce, while unstressed short vowels in closed syllables do not. This is exemplified in the following four examples in (2) taken from Table 3:

(2) a. pózīt – pəzífən
b. rápīd – rəpídətīj
c. hóstajl – həstílətīj
d. káptīv – kaptívətīj

As can be seen in (2), the open syllables in (2ab) allow for reduction. By contrast, the closed syllables (2cd) do not. The word pairs in (2a-d) raise the question as to why closed syllables do not allow for the reduction of short vowels. Burzio (1994: 114-115; cf. Burzio 2007) argues that this distinction is quite natural in the sense that consonants "can in general be

<sup>&</sup>lt;sup>23</sup> One example was found in which /I alternates with [ə] (see appendix A). Nevertheless, it clearly does not follow the general pattern and can thus be considered an exception.

articulated only as transitions between openings and closures of the vocal tract, hence in this sense needing a vocalic 'support'" (1994: 114-115). Consequently, the reduction of short vowels in a closed syllable (VC\$C) would reduce the vocalic support of the first consonant to some degree and hence is not permitted. Conversely, open syllables do not face any similar restrictions. The vowel in a syllable sequence such as V\$CV can be reduced as the consonant in the onset of the following syllable receives the needed support from the nucleus it precedes (1994: 115).<sup>24</sup>

The distinction between open and closed syllables is a useful generalisation to capture the facts found in SSB. However, there is a set of consonants that, when in coda position, allows for vowel reduction. Consider the examples in (3) below:

- (3) a. segmént ségməntéjfən
  - b. kómpleks kəmpléksətij
  - c. méntəl mentálətıj
  - d. kənfránt kónfrantéjjən

The data in (3ab) suggest that sonorants in coda position do not necessarily block vowel reduction. It is generally assumed that sonorants exhibit high sonority (cf. Clements 2009; Martinez-Gil 2001) and can thus more easily stand on their own. Put differently, they do not seem to require vocalic support to the same extent as other consonants (Burzio 1994: 115). This is further supported by the fact that sonorants in English can also occur in the nuclear position in syllables, i.e. they can be syllabic (e.g. RP [kə:tŋ]).<sup>25</sup> What needs to be pointed out here is that while reduction is possible in syllables closed by a sonorant, it is by no means obligatory (1994: 116). This is exemplified in the word pairs in (3cd), which, as opposed to (3ab), show unreduced vowels in unstressed position. Consequently, syllables closed by sonorants may be reduced by some speakers, but not by all. Possible factors influencing whether such syllables reduce or not are, for instance, word-frequency and semantic transparency (to be discussed in later sections of this thesis).

What remains to be briefly discussed is the reduction of syllables closed by *s*. While Burzio (1994: 115) argues that syllables closed by *s* reduce as well, the data set collected for

<sup>&</sup>lt;sup>24</sup> As noted by Burzio (1994: 115), this view is supported by cross-linguistic evidence from French (cf. Jacobs 1989) and Palestinian Arabic (Halle & Kenstowicz 1991).

<sup>&</sup>lt;sup>25</sup> It should be noted though that in SSB, syllabic consonants are less frequently used than in RP. See section 4.2.2 above.

the present study does not support this generalisation. Rather, *s* quite generally seems to block reduction similar to any other consonant in coda position, e.g. /h\_ostajl/ - /h\_ostiletij/. The non-reduction of /ɔ/ in *hostility* suggests that *s* is syllabified as part of the coda of the first syllable. However, it should be noted that *s* may in principle also be considered to belong to the onset of the second syllable. Consequently, there are two possibly syllable structures, i.e. *hos\$tility* vs. *ho\$stility*. Fudge (1984: 198) provides independent evidence for the former syllable structure, namely that open syllables tend to be more frequent if a word has a prefix (e.g. *a*- in *astronomy*). It follows that words such as *hostility* or *festivity*, which do not show any prefixation, are most probably syllabified as /hos\$tilityi and /fes\$tivetij/. This is further substantiated by Goad (2012), who argues that s-clusters should be cross-linguistically analysed as coda+onset. This is the approach followed in the present thesis, since syllabifying words such as *hostility* and *festivity* according to the Maximal Onset Principle (MOP), i.e. *ho\$stility* and *festivity*, cannot account for the regular non-reduction of short vowels preceding s-clusters. The following subsection considers a special case of reduction, viz. the [ə] and [1] alternation found for /ɛ/.

# 6.2.1.3. $/\epsilon/: [a]$ and [I] alternation

While the distinction between open and closed syllables proves to be a useful generalisation, not all aspects of the data can be explained solely in that way. This becomes evident with respect to the vowel  $\epsilon$ , which reduces in two different ways, viz. to [ə] and [I] respectively (cf. Table 3 in section 6.2.1.1). A close examination of the data reveals a clear distribution of both reduced vowels. Consider the word pairs in (4), which illustrate this phenomenon:

- (4) a.  $\acute{e}$ dıt ıdífən
  - b. trépid trípiditij
  - c. əpélət ápəléjʃən
  - d. segmént ségməntéjfən

These word pairs clearly indicate that [I] predominantly occurs in word-initial open syllables (examples (4ab)), while [ $\vartheta$ ] occurs elsewhere ((4cd)). Even though only three instances where found in which  $/\varepsilon$ / reduces to [ $\vartheta$ ], no particular phonotactic context can be identified in those examples that might trigger its occurrence. This stands in stark contrast to the distribution found for [I], which are almost exclusively found in word-initial position. This claim is further substantiated by the fact that the strengthening of [I], i.e. its stressing, quite regularly

results in the vowel  $\langle \epsilon \rangle$  in word-initial syllables. For instance, if, as a consequence of affixation, stress falls on the first syllable of *explain* /ikspléjn/, [I] changes its quality and becomes [ $\epsilon$ ] as in *explanation* /éksplənéjʃən/. Similar changes can be found in the word pairs *expect* – *expectation, reveal* – *revelation* or *present* – *presentation* (cf. appendix A in section 11). By and large, [I] was not found word-medially in unstressed syllables. Thus, the word pairs *perpetuate* /pəpétfuwejt/ – *perpetuity* /pś:prtjúwətij/ and allege /əlédʒ/ – allegation / álıgéjʃən/ may be considered exceptions, since on the basis of the collected data, no definite explanation can be given for the occurrence of [I] word-medially. A possible reason may be the nature of the following consonant, i.e. /tʃ/ and /g/, which may trigger the vowel [I]. However, a larger amount of data would be needed to ascertain this hypothesis. The next section moves to some of the exceptions identified in the data set.

# 6.2.1.4. Exceptions

Exceptions can be found in both unstressed open and closed syllables. A thorough discussion of all exceptions would go beyond the scope of this thesis. Therefore, only three examples will be given in (5) below:

- (5) a. ákses əksésəbílətij
  - b. métəl metálık
    - c. pásiv pasívətij

What all the exceptions have in common is that they do not follow the otherwise quite regular patterns of reduction in English. (5a) reduces even though the syllable is clearly closed by the velar stop [k]. The examples in (5b-c) both show open syllables in unstressed position that do not reduce. It should be emphasised that no systematic distribution emerges from the exceptions found in the data set. However, some possible explanations will be discussed in the following.

Quite generally, exceptions do not pose a major problem to the approach followed in the present thesis, as one of its corner stones is the emphasis of language use. As already discussed in section 2.2, each linguistic unit is considered part of the mental grammar of native speakers. In other words, CG is a maximalist, bottom-up and non-reductionist approach. Consequently, like any other conventionalised expression in a language, exceptions are included in the grammar as concrete instantiations (see section 2.1.3). Moreover, vowel reduction is a gradual phenomenon influenced by many factors among which frequency may
take a pivotal role. Highly frequent words are commonly assumed to resist regularisation and consequently often reduce (Bybee 2001: 28) and may thus exhibit reduced vowels even though the phonotactic context would normally block reduction. At the same time, as will be shown, items showing a low token frequency tend to show full vowels in unstressed syllables. For instance, the derivatives given in (5bc) above both show a rather low frequency, which might explain the non-reduction of the first syllable. Furthermore, this may also be related to issues of semantic transparency, i.e. how transparent the meanings of two related words are. How such factors may be related to the unexpected retention or reduction of a particular full vowel will be explored in later sections of this thesis. Having discussed short vowels, the thesis will now move on to long monophthongs.

# 6.2.2. Long monophthongs

#### 6.2.2.1. The data

The reduction of the six long monophthongs in SSB is illustrated in Table 4 below. Similar to the examples in Table 3 above, the first column of each vowel gives the context in which reduction occurs, the second the context in which the full vowel is retained:

$I: \rightarrow 9$	E	$\mathfrak{e}:  ightarrow \mathfrak{d}$	33
əpí: antí:rɪjə		dəklé:	έːrɪst
ápərí∫ən	ánturijórətij	dékləréj∫ən	εːrístık
$\mathfrak{a}: \rightarrow \mathfrak{d}$	a:	$\mathfrak{H} \to \mathfrak{H}$	ə:
páːtıkəl	ımbáːk	kənsə́:v	á:bən
pətíkjələ	émba:kéj∫ən	kónsəvέj∫ən	əːbánətɪj
ədvá:ntıdz	á:tıst	pá:fikt	ıkstə́:nəl
ádvəntéjdzəs	a:tístik	pəfék∫ən	ékstə:nálətıj
0: → 9	0:	$\Theta$ ; $\rightarrow$ $\vartheta$	
ɪnfóːm	kó:z	bjéːrəw	
ínfəmέj∫ən	ko:zέj∫ən	bjərókrəsıj	
ıksplóĭ	ό:θə	әत्दर्भ:	
ékspləréj∫ən	o:θórətıj	ádʒəréj∫ən	

 Table 4 Reduction of long monophthongs

As can be seen in the examples given in Table 4, long monophthongs in SSB reduce to [a] in unstressed syllables. Thus, with respect to long monophthongs, only prominence-reducing, or centripetal, vowel reduction can be found. For the final vowel (a:/, no context could be found)

identified in which reduction does not occur. Based on the data set, it may be stipulated that /  $\Theta$ :/ reduces whenever stress is lost. Moreover, it should be emphasised that for the monophthongs /I:/, / $\epsilon$ :/ and / $\Theta$ :/, only a limited number of word pairs was found in the data collection process. Their restricted occurrence may be due to the fact that these vowels were originally pronounced as diphthongs in RP but have undergone monophthongisation in SSB.

As was mentioned with respect to short vowels, lack of stress is not the only factor governing vowel quality. However, the picture presented by long monophthongs is somewhat more complex than that of short vowels. In the literature, there are different perspectives on the behaviour of long monophthongs in unstressed positions. While Burzio (2007: 162) argues that long vowels are immune to reduction due to their longer duration, Crosswhite (2001: 205) notes that long vowels reduce via laxing. Considering the data presented in Table 4, it can clearly be seen that long monophthongs often do undergo changes in unstressed syllables. Moreover, long vowels can be found in both stressed and unstressed syllables. Unlike with short vowels, the changes in the vowel quality of long monophthongs cannot be attributed to syllable structure. Rather, a number of other factors such as foot structure, frequency effects and semantic transparency may influence whether a given long monophthong reduces or not. The most important of these factors, viz. foot structure, will be discussed in the following subsection.

# 6.2.2.2. Reduction of long monophthongs and foot structure

In many cases, reduction of long monophthongs in SSB seems to be conditioned by foot structure. In principle, unfooted syllables do not reduce in SSB. This is to say that a word-initial long vowel which loses stress as a result of affixation is typically not reduced if it is directly followed by a stressed syllable, i.e. the head of the following foot. Put differently, the word-initial syllable is not integrated into the foot structure of the word and thus retains its full quality. Conversely, if a formerly stressed syllable becomes an unstressed syllable in a foot, it is reduced. Some examples are given in (6) below:

(6)	a. áːtɪst	_	a:(tístik)
-----	-----------	---	------------

b. kó:z – k	to:(zέj∫ən)
-------------	-------------

- c. əbzə́:v − (ɔ́bzə)(vɛ́jʃən)
- d. dəklé: (déklə)(réjʃən)

The word pairs in (6) illustrate the sensitivity of vowel quality to foot structure. This observation is congruent with the results of a cross-linguistic study conducted by Wedel et al. (2019) that suggests that segments at the beginning of words contribute a larger amount of information on word identity than segments occurring towards the end of words. What follows is that due to their greater importance in word identification word-initial segments are less likely to undergo reduction processes than word-final segments (2019: 245).

The latter two examples in (6) above indicate that long monophthongs reduce if they are integrated into the foot structure of a word. In the derivatives in (6cd), secondary stress is placed on the first syllable of the respective derivative.<sup>26</sup> Wenszky (2004: 11) argues that this is due to the alternating rhythm of English, which disfavours more than two adjacent unstressed syllables (particularly in word-initial position). For example, since main stress is moved from the second syllable in observe /əbzə́:v/ to the affix /-éjfən/ in /ɔ́bzəvéjfən/, secondary stress is placed on the first syllable to avoid two consecutive unstressed syllables. The secondarily stressed syllable and any unstressed syllable following it constitute feet by themselves. The data collected suggest that if such a foot occurs, the unstressed syllable undergoes reduction to [ə]. As can be seen in the derivatives (6cd), each word is prosodically structured into two feet, the first of which is weak, i.e. secondarily stressed, and the second of which is strong, i.e. primarily stressed. Since SSB exhibits foot-based vowel reduction, unstressed syllables in feet reduce. However, not all the word pairs collected can be explained that way. As already mentioned, the reduction of long monophthongs is an interplay of different factors all of which contribute to their behaviour in unstressed positions. These will be looked at in more detail in the following section.

# 6.2.2.3. Other factors influencing (non-)reduction of long vowels

The tendencies outlined above for long monophthongs are valid generalisations made on the basis of the data set. However, these generalisations are by no means as regular as those established for short vowels. Other factors such as the frequency of a particular item or its semantic transparency may notably influence the reducibility of a long monophthong in SSB. Frequency can generally have two seemingly contradictory effects on linguistic patterns. On the one hand, highly frequent items tend to undergo phonetic changes (particularly in the

<sup>&</sup>lt;sup>26</sup> Note that secondary stress is not transcribed with a different symbol in CUBE. The right-most stress is primary, while any stress mark to the left of it necessarily indicates a secondarily stressed syllable.

process of grammaticalisation) more easily (Bybee 2001: 11). On the other, high frequency items also resist regularisation in many cases, as can be observed, for example, in the irregular past tense forms in English (2001: 12). With respect to the data analysed in the present thesis, frequency effects seem to result in the preservation of full vowels in items of low frequency. Compare the examples given in (7). The first two are relatively frequent items, the latter two are not:<sup>27</sup>

- (7) a. dráːmə drə(mátık)
  - b. páːtɪkəl pə(tíkjələ)
  - c. dəfró:d (dífro:)(dɛ́jʃən)
  - d. ıkstá:nəl (ékstə:)(nálətıj)

The items in (7) are all exceptions to the generalisations discussed in the previous section. What can be seen is that the first vowels in the highly frequent items in (7a) and (7b) reduce even though they are not integrated into the foot structure in the derivatives. By contrast, the vowels in (7c) and (7d) do not reduce in unstressed position. A possible reason for the non-reduction is their comparably low frequency. However, no statistical analysis was carried out as part of this thesis. Thus, no definite claims on the effect frequency has on vowel reduction in English can be made at this point. Nevertheless, the data collected suggests a tendency for less frequent items to resist reduction in SSB.

Apart from frequency, semantic transparency may similarly have an effect on whether unstressed long monophthongs reduce. Semantic transparency is generally defined in rather vague terms as "how transparent the end product of a morphological process is with regard to its meaning" (Bell & Schäfer 2016: 158). Put differently, if the meaning of a word can be predicted based on the word-formation processes it was formed by, it is said to be semantically transparent (cf. Plag 2003: 46). Other definitions take meaning-relatedness between two items as the central factor governing semantic transparency (e.g. Zwitserlood 1994). Whatever definition is taken, the effects of semantic transparency on vowel reduction are relatively hard to ascertain. It seems to be the case that highly non-transparent words such as *particular* /pətíkjələ/ are more likely to show reduction than semantically more transparent words, e.g. *embarkation* /ɛ́mba:kėjʃən/, in which the meaning is predictable on the basis of

 $<sup>^{25}</sup>$  The frequencies of the respective items were checked using CELEX. Since the size of the corpus is not immediately clear, no token counts will be given in the thesis. Put differently, the derivatives in (7ab) are frequent compared to those in (7cd). However, no claim as to the overall frequency of the items in English can be made.

*embark.* This, however, may also be attributed to the former's relatively high frequency. Consequently, what becomes evident is that semantic transparency and frequency correlate with each other to some extent. In many cases, the degree of semantic transparency alone does not suffice to explain the (non-)reduction of a given item in the data set. At this point it is simply important to note that many of the exceptions found in the data on long monophthongs may be attributable to factors other than foot structure such as, for example, said semantic transparency. A possible analysis of those effects on vowel reduction will be proposed in section 7.4.

Another factor that may be of importance in the behaviour of unstressed vowels in SSB is the position a particular unit takes in the linguistic system. For example, the derivatives in the word pairs conform /kənfó:m/ - conformation /kónfo:méj[ən/ and confirm / kənfá:m/ - confirmation /kónfəméjfən/ differ with respect to the vowel quality with the former showing no reduction of /o:/ and the latter reduction to [ə]. Since the vowel quality is what differentiates both derivatives from each other, reducing /o:/ in conformation would presumably lead to difficulties in word recognition. As a consequence, it may well be assumed that the full vowel in *conformation* is retained to maintain their difference in the linguistic system. The analysis presented in section 7.4 will propose how frequency effects and semantic transparency may be accounted for in CG. It remains to be emphasised that the aim of the present thesis is not to once and for all solve all issues concerning vowel reduction. Vowel reduction in English is an intricate phenomenon in which an array of different factors interplay in different ways. A thorough treatise of how item frequency, semantic transparency and vowel reduction correlate with each other would go beyond the scope of this thesis. Rather, the thesis aims at outlining a CG framework that principally allows for accounting for phonological phenomena and will show possible ways of explaining exceptions related to frequency effects and semantic transparency in section 7.4. The following section introduces the data on diphthongs considered in the subsequent analysis.

# 6.2.3. Diphthongs

### 6.2.3.1. The data

The final set of vowels not yet introduced is the seven diphthongs found in SSB. The behaviour of diphthongs in vowel reduction is shown in Table 5 below. As with short vowels

and long monophthongs, the last column for each vowel gives the context in which no reduction takes place:

ıj → ə	ıj →ı	ıj	εj → ə	εj → I	εj
kəmpíjt kómpətí∫ən	íjkwəl ıkwólətıj	líjgəl lıjgálətıj	ıkspléjn éksplənéj∫ən	əbstéjn ábstınəns	néjzəl nejzálətıj
əpíjl ápəlέj∫ən	síjkwəns sıkwén∫əl	íjsðījt ījsðétik	éjbəl əbílətıj	pətéjn pəʿːtɪnəns	έj&ənt εj&én∫əl
$aj \rightarrow a$	aj→ı	aj	$\Im M \rightarrow \Im$	әw	
ədmájə ádməréj∫ən	dəzájn dézıgnéj∫ən	fájnəl fajnálətıj	pówlə pəlárətıj	īváwk íjvəwkéj∫ən	
sátajə sátərajz	əblájdz óblıgéjʃən	sájt sajtéj∫ən	əpə́wz ópəzí∫ən	mə́wdəl məwdálətıj	
uw → ə	uw		oj		aw
əkj <del>ú</del> wz ákjəzέj∫ən	br <del>ú</del> wtəl br <del>u</del> wtálətrj		ımplój émplojíj		fáwnd fawndéj∫ən
rıpj <del>ú</del> wt répjətəbəl	ıkskl <del>ú</del> wsıv ékskl <del>u</del> wsívətıj		ıksplójt éksplojtéj∫ən		áwtrejdz awtréjdzəs

Table 5 Reduction of diphthongs

The reduction processes of diphthongs in SSB appears to be somewhat more complex than that of short vowels and long monophthongs. Three out of the seven diphthongs reduce to either [ə] or [I], i.e. exhibit centripetal, i.e. contrast-reducing, and centrifugal, i.e. contrast-enhancing, patterns. Of the remaining four diphthongs, two, i.e. /oj/ and /aw/, do not reduce at all, while the diphthongs /əw/ and /uw/ only reduce to [ə]. A note is due on the vowel / $\varepsilon$ j/. Only two examples were found in which / $\varepsilon$ j/ reduces to [I]. Additionally, both words seem to have been borrowed directly from French (OED 2019: s.v. *pertinence; abstinence*). Based on the data set, it may be said that reduction of / $\varepsilon$ j/ to [I] does not seem to be productive in SSB, but rather is a result from items being borrowed from French directly. Consequently, they will not be considered in the present thesis. The following two subsections will examine possible reasons behind the patterns observable in Table 5.

## 6.2.3.2. Reduction of diphthongs and foot structure

Similar to long monophthongs, the picture presented by the behaviour of diphthongs in unstressed position seems rather intricate and complex. By and large, the generalisations outlined for long monophthongs hold for most diphthongs as well. For the vowels /aj  $\Rightarrow w \pm w$ /, foot structure appears to be one of the factors conditioning reduction. Consider the word pairs in (8) below:

(8) a. əpówz – (ópə)zífən
b. ımpláj – (ímplı)kéjfən
c. brúwtəl – bruw(tálətij)
d. mówdəl – məwdálətij

The examples in (8a) and (8b) illustrate that SSB exhibits a tendency for reducing unstressed diphthongs if they are integrated into the foot structure of the derivative, i.e. if, together with a secondarily stressed syllable, they form a weak foot. The last two items in (8) show that unfooted initial diphthongs tend to retain their full vowel quality. The remaining four diphthongs exhibit different behavioural patterns in unstressed syllables. Unlike the previous three diphthongs, /ij/ reduces more or less regularly in footed and unfooted syllables. The vowel / $\varepsilon$ j/ takes a similar position in the diphthongal system of SSB and reduces irrespective of foot structure. Nevertheless, exceptions to these generalisations do exist as well and will be discussed in section 6.2.3.3 below. At this point it suffices to note that for many of the counterexamples, frequency effects can offer a plausible explanation. Moreover, the data in Table 5 also indicate that /oj aw/ do not reduce at all. A possible reason will be suggested below.

While foot structure is without doubt one of the conditioning factors of the reduction of diphthongs, it does not explain the [ə]/[I] alternation found in the data on the vowels /IJ/ and /aJ/. Consider the word pairs given in (9) below. (9a-d) illustrate the patterns of /IJ/ and (9e-h) those of /aJ/:

(9)	a. íjkwəl	-	ı(kwʻələtij)	e. impláj)	_	(ímplı)(kέj∫ən)
	b. skíjmə	_	skı(mátık)	f. fájnajt	_	(ínfinət)
	c. kəmpíjt	_	kóm(pətí∫ən)	g. ədmájə	_	(ádmə)(rέj∫ən)
	d. rīvíjl	_	révə(lέj∫ən)	h. sátajə	_	(sátərajz)

The data in (9a-d) clearly show that /ij reduces to [1] in word-initial syllables, whereas [2] occurs in all other footed contexts. The data set exhibits one exception to that pattern, viz. the word pair repeat /rɪpíjt/ - repetition /répɪtíʃən/ in which [1] is found in footed, non-initial position. The generalisations that can be established for /aj/ differ. In principle, /aj/ only reduces if it is integrated into the foot structure of a word. With respect to the data in (9e-h), it can be observed that reduction to [1] generally does not follow a clear pattern. Consequently, [1] may be regarded as the elsewhere case. By contrast, the occurrence of [2] seems to be restricted. In all word pairs collected for the reduction pattern  $/aj/ \rightarrow [a]$ , the base word of the derivative ends in the triphthong /ajə/ resulting from the vocalisation of /r/ in word-final position (as typical for non-rhotic varieties of English). When stress shifts away from /aj/ as a result of affixation, the vowel reduces to [a], which is directly followed by /r/. The occurrence of /r/ in those contexts can be treated as a word-medial sandhi phenomenon linking the reduced vowel and a suffix beginning with another vowel. Consequently, [ə] only occurs in a phonotactically restricted environment, viz. if it directly precedes /r/-sandhi in the derivative. However, it should be mentioned that the data on diphthongs considered here is limited to some extent. A larger data set would yield more conclusive evidence for the generalisations established. Nevertheless, clear patterns for the reduction of diphthongs do emerge. The subsequent section moves to the discussion of examples that deviate from the general patterns discussed here.

6.2.3.3. Other factors influencing (non-)reduction of diphthongs

Similar to long monophthongs, frequency effects may also significantly influence whether a given diphthong reduces or not. Atypical reduction patterns can be found for all diphthongs in the data set. A particularly evident case in which frequency influences the quality of diphthongs concerns the reduction of  $/\epsilon j/$ . The only two instances found in which the diphthong retains its original vowel quality in the derivative are two low frequency items, viz. *nasality* /nɛjzálətij/ and *agential* /ɛjdʒɛ́nʃəl/. In all other cases, /ɛj/ regularly reduces to either [ə]. Another set of exceptions to the tendencies identified can be found in the word pairs of /əw/. It appears to be the case that in a number of derivatives /əw/ reduces to [ə] in unstressed word-initial, and thus unfooted, position, such as *photography* /fə(tɔ́grəfij)/ and *momentous* /mə(mɛ́ntəs)/. These two examples show reduced vowels in a context which usually does not allow reduction, i.e. unfooted syllables. While frequency effects may

account for some of the cases found in the data set (*photography* is arguably relatively frequent), others are more likely to be the result of a complex interplay between frequency, semantic transparency and other factors. The derivative *momentous*, for instance, can be considered semantically intransparent. Put differently, its meaning cannot be arrived at by simply resorting to the meaning of the base word. For now, it should only be mentioned that semantically intransparent items may allow for reduction in unfooted syllables. A possible analysis taking semantic transparency into consideration and possible reasons for this behaviour will be presented in section 7.4.

An additional question raised by the data set is the non-reduction of /oj aw/ raises the. Even though the data collected is limited, it does not suggest any particular phonotactic reasons which would justify the retention of their full quality. It may be argued that the diphthong /aw/ was only found in word-initial, unfooted position and accordingly is not reduced, e.g. outrageous /awtréjdʒəs/. In other words, /aw/ was not found in any context in which it would be expected to undergo reduction. However, a potentially more powerful explanation for the non-reduction of /oj aw/ is the fact that both diphthongs are usually analysed as consisting of three morae (Hammond 1999: 205). A mora is typically defined as "a unit of quantity for syllables" (1999: 40). While mora-based generalisations have deliberately not been employed in the generalisations established in the previous sections as they do not provide any advantages to the proposed analysis, they may shed light on the behaviour of /oj/ and /aw/ in unstressed position. In moraic theory, it is generally accepted that lax vowels are monomoraic, while tense vowels and all other diphthongs are bimoraic (1999: 205). Moreover, reduced vowels are taken to be nonmoraic in nature. It follows that vowel reduction can then be described as a loss of morae in unstressed position (Hammond 1997: 3). What seems to be the case for the diphthongs /oj aw/ is that, due to their trimoraic character, they do not undergo any reduction. A loss of one mora results in a bimoraic and therefore still long vowel.<sup>28</sup> However, while useful in the analysis of other languages such as Japanese, the mora does not appear to be of immediate relevance for the phonology of English. Starting from a usage-based perspective, speakers are assumed to form schemas over

<sup>&</sup>lt;sup>28</sup> It should just be mentioned at this point that the theory outlined in this thesis does not proscribe morae. Moreover, the proposed moraic analysis of /oj aw/ is a tentative one which requires more detailed analysis. Moraic theory is a popular approach to linguistic phenomena in many different languages and cannot be adequately discussed in this thesis. See Hyman (1985) for a comprehensive treatise of the theory.

the behaviour of /oj/ and /aw/ that specify their non-reduction. Consequently, none of the schemas developed in the analysis will resort to the concept of the mora.

A number of atypical patterns still remain to be discussed. However, not all exceptions found in the data set can be considered in this section. While it should be emphasised again that such exceptions do not pose a problem for an analysis couched in CG, it is important to note that in principle, word frequency and semantic transparency provide means of accounting for most exceptions. Moreover, according to CG's commitment to usage-based approaches, each conventionalised linguistic unit is contained in the grammar. Thus, even though such exceptions do not follow the higher-level schemas for vowel reduction in English, they are stored as concrete instantiations, i.e. highly specific schemas, in the speakers' mental grammar. How exceptions can be treated in a CG framework will be explored in section 7.4 below. Since the reduction patterns discussed in this section are relatively complex, the following section briefly summarises the generalisations established so far.

### 6.2.4. Reduction in SSB: a brief summary

For ease of reference, the (regular) vowel reduction patterns are briefly summarised for each set of vowels in Table 6 below. Additionally, the relevant conditioning factor is indicated: **Table 6** Reduction patterns in SSB summarised

		Conditioning factor		
Short vowels				
V	$\rightarrow$ [ə]	in open syllables		
/ɛ/	$\rightarrow$ [I]	in word-initial open syllables		
Long monophthongs				
VV	$\rightarrow$ [ə]	if footed		
Diphthongs				
VV	$\rightarrow$ [ə]	if footed		
ıj	$\rightarrow [I]$	in word-initial, unfooted syllables		
	→ [9]	elsewhere		
εj	$\rightarrow$ [ə]	irrespective of foot structure		

aj	$\rightarrow$ [ə]	if footed
	$\rightarrow$ [I]	in r-sandhi contexts
oj		does not reduce
aw		does not reduce

Having explored the major tendencies for and possible reasons behind the reduction patterns in SSB, the thesis now will move on to the analysis of the phenomenon.

#### 7. A CG analysis of vowel reduction in SSB

In this section, an analysis of vowel reduction in SSB couched in CG will be proposed. Since the motivating factors behind the reduction of short vowels, long monophthongs and diphthongs are disparate, they will be dealt with in separate sections. Section 7.1 will present an analysis of short vowel reduction, section 7.2 will consider long monophthongs and section 7.3 will examine diphthongs in more detail. The final section, 7.4, will show how frequency effects and semantic transparency may be accounted for in the framework.

# 7.1. Reduction of short vowels

### 7.1.1. The schemas

The reduction of short vowels in SSB critically hinges on the nature of the syllable it occurs in, viz. whether it is open or closed. As was outlined in the previous sections, CG emphasises the importance of the formation of cognitive schemas based on speakers' experience with language. Consequently, a number of schemas emerge from exposure to utterances. Considering that vowel reduction is conditioned by the structure of the syllable, it becomes evident that two different types of schemas are needed in the analysis. These include, on the one hand, schemas capturing syllabification in English and, on the other, schemas generalising over changes in vowel quality. Furthermore, it should be emphasised once again hat schemas do not exist in an empty space in the mental grammar but rather form networklike structures by interacting and competing with each other. Vowel quality and syllable structure are closely related to each other and consequently form such a network of schemas themselves. The diagrams illustrating the interaction of those schemas presented below are relatively complex. In order to make the presentation of the analysis more accessible, each type of schema will be considered in isolation first. Three different types of syllables relevant with respect to the reduction of short vowels can be identified (cf. section 6.2.1.2). The first two types, i.e. open and closed syllables, have already been mentioned above. The third type can be found in words exhibiting s-clusters. As a consequence of exposure to language, speakers may be expected to form systematic relationships between sequences of sounds and syllable structure. These relationships can be captured in second-order schemas, which relate schematic sound sequences and possible (schematic) syllable structures. The diagram in Figure 6 below shows three second-order schemas for the three types of syllables found in the data and their respective first-order schemas:



Figure 6 Syllable schemas<sup>29</sup>

Each second-order schema in Figure 6 is built from the two first-order schemas to the left of it. Since language users encounter different types of sound sequences, they can form abstract schemas generalising over them, i.e. they can establish schemas capturing particular sequences of consonants and vowels. This is shown in each of the left-most first-order schemas in Figure 6. Moreover, while the first-order schema in 6a and 6b show highly schematic sequences of sounds, the schema in 6c is more specific, viz. it specifies the type of consonant, i.e. /s/, in the sound sequence. Additionally, possible syllable structures in English are captured by the right first-order schemas, viz. open syllables (in 6a), closed syllables (in

<sup>&</sup>lt;sup>29</sup> The dotted boxes around the second-order schemas are meant to illustrate that syllable schemas are high-level, i.e. very abstract, schemas.

6b) and syllables closed by /s/ (in 6c).<sup>30</sup> Considering the non-reduction of vowels immediately preceding an s-cluster (cf. section 6.2.1.2), /s/ should be analysed as part of the coda and not of a complex onset (cf. Goad 2012). Note, however, that the first-order schemas do not say anything about syllabification yet. They only capture generalisations over utterances, but do not spell out how they are related.

The second-order schemas in Figure 6 capture systematic relationships between firstorder schemas or, put differently, between sound sequences and possible syllable structures. Consequently, speakers may hold phonological schemas in their mental grammar that specify that the syllable boundary of a sequence such as (C)V is typically placed after the vowel, i.e. (C)V\$ (cf. 6a in Figure 6). In other words, it can be assumed that speakers recognise systematic relationships between the sound sequence CV and open syllables CV\$. Accordingly, they can generalise over this relationship and form second-order schemas. In a similar fashion, speakers may also form second-order schemas for closed syllables as shown in 6b above. However, it should be mentioned that (C)VC\$ is not the only way in which a sequence such as (C)VCCV can be syllabified. Rather, depending on the type of consonant cluster involved, other syllable structures are possible as well. This, however, is not an issue for the theory outlined in this thesis, since specific schemas take priority over more general schemas (cf. conceptual overlap; Langacker 1999). Put differently, low-level schemas for particular consonant clusters and their syllabification are abstracted from utterances as well and may then cover for more specific cases of syllabification. The final second-order schema in 6c generalises over the syllabification of s-clusters. It states that a sequence of sounds that contains an s-cluster SC is typically separated into coda+onset (cf. Goad 2012).

In addition to syllable schemas, the analysis presented below would be incomplete without establishing schemas capturing the changes in vowel quality. As a general rule, all vowels in open syllables reduce to [ə] when stress moves from the syllable with the exception of  $\epsilon$ , which reduces to either [ə] or [I]. Furthermore, vowels in syllables closed by a sonorant may reduce as well in SSB. Similar to the relations between sound-sequences and syllable structures, speakers may also capture the relations between stressed and unstressed vowels in

<sup>&</sup>lt;sup>30</sup> It should be emphasised that only the syllable structures encountered in the data set are given as schemas here. The aim of this thesis is not to develop a framework of syllabification in CG. Rather, syllable schemas are only used where needed to account for vowel reduction. Moreover, no analysis will be presented for the non-reduction of syllables closed by /s/. It is nevertheless given in Figure 7c in order to account for the syllabification of words such as *festivity* or *hostility*.

second-order schemas. Figure 7 below introduces the second-order schemas and the firstorder schemas they are constructed from:



Figure 7 Reduction schemas for short vowels

The reduction patterns identified in section 6.2.1 above are presented in schematic terms in Figure 7 above. It should be emphasised again that first-order schemas are merely generalisations over actually occurring utterances. They do not show how they are connected to each other in the mental grammar of speakers. These relations are indicated in the secondorder schemas in Figure 7. The second-order schemas in 7a and 7b specify that any mid or low vowel, i.e.  $|a \in \mathfrak{I} \wedge \Lambda|$ , reduces to  $[\mathfrak{I}]$  when stress is lost. Figure 7a can be considered the elsewhere case, since it does not specify any context. By contrast, Figure 7b gives the phonotactic context of reduction, viz. syllables closed by sonorants. The two schemas in Figure 7c and 7d capture the behaviour of the vowel  $\epsilon$ / in stressed and unstressed position. As can be seen, both define a concrete environment in which reduction to [1] occurs, viz. in word-initial syllables. Moreover, Figure 7d covers the reduction of closed syllables exhibiting a sonorant in coda position. The schemas for reduction to [1] are necessarily more specific than the schemas established for mid-low vowels in general. This has important consequences for the analysis presented below, since more specific schemas add additional activation value to their respective candidate expressions (cf. conceptual overlap). The second-order schema in Figure 7e captures the non-reduction of unstressed vowels.

Moreover, note that none of the schemas in Figure 7 specify that reduction only occurs in open syllables. This is captured by the syllable structure schemas presented in Figure 6. The schemas established thus far in isolation suffice to analyse the reduction of short vowels within the framework developed in the previous chapters. The following section will bring them together and present the analysis.

# 7.1.2. Analysis

Before discussing the analysis, a number of technical comments are needed to make the presentation more comprehensible. Since the analysis presented in this section is rather complex, it will only be shown on the example of selected words. It is important to understand, however, that the analysis applies to all the words collected in the data set. Thus, in principle any word following the same pattern can be substituted for the items used below. Moreover, to make the graphical representation of the analysis easier, schemas that do not apply to the example discussed will not be given in the diagram. For example, if reduction in open syllables of mid-low vowels is discussed, the schemas for reduction of vowels closed by sonorants will be omitted. Nevertheless, in theory all schemas are assumed to interact with each other. This section will first discuss reduction in open syllables and then move on to the reduction of  $/\epsilon$  and the reduction of syllables closed by sonorants.

### 7.1.2.1. Reduction in open syllables

The simplest case of vowel reduction can be found in open syllables. In order to account for this, several schemas are needed. First, the syllabification schema introduced in Figure 6a above is necessary to make sure vowels only reduce in open syllables. Moreover, two reduction schemas, viz. the ones in Figures 7a and 7e, are also required. As a connectionist model, CG assumes that the schemas in a speaker's mental grammar form complex networks. Consequently, the analysis proposed critically hinges on categorisation relationships between those schemas, which are indicated by arrows. In other words, an arrow between a schema A and a schema B indicates that B elaborates, i.e. is compatible with, A (cf. categorisation relationships in section 2.2). The diagram in (13) shows the analysis of open syllable reduction on the example of the word pair *posit* /pźztt/ – *position* /pǝzíʃǝn/:

#### (13) Reduction in open syllables



The network-like structure briefly mentioned above is graphically represented in (13). The schemas in (13a) and (13b) interact with each other, i.e. (13a) specifies syllable structure, while (13b) captures the quality of mid-low vowels if stress is moved. Note that the bold box around the two schemas indicates that open syllables and vowel reduction are closely related. Put differently, it captures the tendency observed in SSB that open syllables typically reduce, while closed syllables do not. In contrast, the schemas in (13b) and (13c) compete against each other. What this means is that the conventionalised linguistic unit (13d) in the mental grammar of a speaker cannot be categorised by both. Either the vowel in question is reduced or it is not. Moreover, note (13b)'s lower graphical position in the diagram, which indicates its higher degree of specificity, i.e. its closer cognitive distance to the candidate expression. Furthermore, the linguistic unit in (13d) relates the phonological poles of the two words *posit* and *position*. Since it specifies the concrete lexical entries in their phonological form, the two first-order schemas can be regarded as conventional linguistic units. The lower box is given in bold to indicate the additional activation value that is added to a candidate expression by

linguistic units. Note that (13d) instantiates or, in other words, is categorised by both (13a) and (13b).

The candidate expressions actualised by the schemas contained in a speaker's mental grammar are given in (13e) and (13f). The expression in (13e) shows a reduced vowel in the first syllable. It is categorised by the syllable schema in (13a) and the reduction schema in (13b). Moreover, (13e) is also categorised by the conventional linguistic expression given in bold in (13d). (13f) presents a candidate expression, the first vowel of which, while showing the correct syllable structure, is not reduced. The candidate in (13f) is categorised by the syllable schema in (13a) and reduction schema in (13c). It should be stressed once again that the number of candidates is theoretically infinite. In other words, it may well be that other expressions, such as wrongly syllabified candidates, are actualised in the process of retrieving the correct form. In the analyses to follow, not all logically possible candidate expressions will be given. Rather, in most cases, two or three alternatives suffices to show how schemas interact to select the correct winner. As a final note on candidate expressions, it is crucial to understand that the sanctioning of the candidates in (13) is not arbitrary, but rather emerges from the well-formedness principle *access* (cf. section 3.2.2).

In order to solve the competition between (13e) and (13f), the speaker compares the candidate expressions to the schemas they hold in their mental grammar. The well-formedness principles provide a means of "[determining] the well-formedness of [a candidate]" (Kumashiro 2000: 24) and are consequently necessary to select the correct winner. The calculation of the total activation value of each candidate is based on the second well-formedness principle *activation* (cf. principle (1b) in section 3.2.2). According to (1b), each categorising unit, i.e. schema, which an expression is categorised by, adds to the final activation value of the candidate. It is generally assumed that the closer the cognitive distance of a candidate to the schema it instantiates, the higher the activation value obtained (cf. Kumashiro 2000: 25). Close cognitive distance positively correlates with the degree of conceptual overlap. Put differently, if a candidate conceptually overlaps with a schema to a high degree, the cognitive distance between them is relatively close. Consequently, candidates in close cognitive distance to their categorising unit, i.e. those which exhibit high conceptual overlap, obtain a higher amount of activation value than candidates showing a greater distance to their respective schemas.

The competition between the candidates in (13e) and (13f) can be resolved as follows. Considering the diagram presented in (13), it can be seen that the schema in (13e) is categorised by two schemas, i.e. (13a) and (13b). The schema in (13b) is relatively specific as it determines the type of vowel, viz. mid-low, that it applies to. Moreover, the syllable schema (13a) and the reduction schema (13b) are closely related to each other, which is represented by the bold box.<sup>31</sup> It follows that the candidate expression obtains additional activation value from both the close relation between (13ab) and from conceptually overlapping with (13b) to a higher degree. Additionally, the schema in (13d) also contains the conventional linguistic unit *position* [pəzífən] in bold, which further increases the activation value of the candidate in (13e).<sup>32</sup> By contrast, the expression in (13f) is categorised by two schemas, i.e. (13a) and (13c) only. Moreover, since (13c) is less specific than (13b), the cognitive distance to the candidate is greater. Hence, the total activation of (13f) can reasonably be considered lower than that of (13e). The principle in (1c) (section 3.2.2), i.e. uniqueness, ensures that only one candidate, namely the expression having obtained the highest activation value, is selected as the winner of the competition. Thus, the model developed in this thesis correctly predicts the winning candidate, viz. (13e) in bold. The other candidate expression (13f) is deactivated as a result of (1c).

The last principle (1d), i.e. *well-formedness*, deserves a number of comments at this point. It should be noted that, in principle, it is not necessary in resolving the competition. Rather, it makes predictions about the extent to which a candidate expression can be considered well-formed. In other words, a candidate exhibiting a high amount of total activation value is more well-formed than a candidate showing a low amount of activation value regardless of it being selected as the winning candidate. This has interesting implications as to the historical development of languages. It was briefly mentioned above that the *uniqueness* principle is not absolute. In the process of historical change, two candidate expressions may be activated and realised by speakers in utterances. In such cases, the model developed in this thesis would predict that, if no other factors such as frequency or semantic transparency intervene, the candidate with the higher degree of well-formedness,

<sup>&</sup>lt;sup>31</sup> Even though in principle possible, the bold box will not be treated as a schema in its own right. However, it should be mentioned that this relationship influences the analysis considerably. Thus, an arrow can be found ranging from the bold box to the candidate expression in (13e).

<sup>&</sup>lt;sup>32</sup> It should be emphasised that the correct winner is predicted regardless of the conventional linguistic unit. It is given only for reasons of clarity and will be left out in the subsequent analyses.

i.e. the higher amount of total activation value, eventually wins the competition. What follows is that the other candidate, which at the beginning of the competition is used by speakers as well, is predicted to be lost over time. Having discussed the most prototypical cases of reduction, this thesis will consider more specific patterns of short vowel reduction in the following two subsections, starting with the reduction of  $/\epsilon/$ .

# 7.1.2.2. The reduction of $\epsilon$ / in open syllables

The analysis presented in (13) above covers the facts of all short vowels found in SSB. However, it does not capture the reduction of  $\epsilon$ / in word-initial open syllables, in which it reduces to [1] rather than [ə]. Consider the diagram in (13) once again. It can be seen that it does not have any means of ensuring correct predictions for word pairs such as *edit* [ $\epsilon$ dit] – *edition* [Idíʃən]. To solve this problem, a new schema generalising over the vowel  $\epsilon$ / in wordinitial syllables may be established (cf. Figure 7c in section 7.1.1). The diagram in (14) below illustrates the analysis on the example of *edit* / $\epsilon$ dit/– *edition* /Idíʃən/:

(14) Reduction of  $/\epsilon/$ 



As can be seen, the diagram in (14) is similar to the analysis presented in (13). The first three second-order schemas, i.e. (14a-c), have already been employed in the previous analysis. In order to make correct predictions about the grammar of SSB, however, the schema in (14d) is necessary. Categorisation relationships are again indicated by arrows ranging from one

second-order schema to another second-order schema or candidate expression. Moreover, the bold box around the schemas in (14a), (14b) and (14d) indicates that vowel reduction and open syllables are closely related. Note further that no conventional linguistic unit is given in (14). While it can reasonably be assumed that such a unit exits, it is not needed to resolve the competition between candidates.

The candidate expressions, which are sanctioned by the categorising units in (14), are given in (14e-g). The candidate in (14e) instantiates the general reduction schemas for open syllables in English, viz. the syllabification schema in (14a) and the reduction schema in (14b). Thus, it is actualised with the reduced vowel [ə] in the first syllable. The second expression in (14f) is categorised by the syllable schema (14a) and the reduction schema in (14d). Consequently, it is sanctioned with the vowel [ɪ] in word-initial position. The final candidate in (14g) instantiates the syllable schema in (14a) and the reduction schema in (14d). However, although correctly syllabified, it does not show a reduced vowel in the first syllable. It should be noted that the reduction schemas in (14b), (14c) and (14d) stand in competition to each other. However, the schema in (14d) is most specific since it specifies the context in which reduction to [I] occurs. Moreover, what needs to be emphasised again is that the set of candidates is in theory infinite. In principle, other schemas could be added to those given in (14), which then would give rise to further candidates. For reasons of readability, however, only the schemas pertinent to the analysis presented are illustrated graphically.

Selecting the winning candidate out of the expressions in (14e-g) is handled by the well-formedness principles introduced in section 3.2.2. The sanctioning of the candidate expressions is motivated by principle (1a), i.e. *access*. Principle (1b), i.e. *activation,* considers the sum of all activation values obtained from the schemas in a speaker's mental grammar. Looking at the diagram in (14), it can be seen that the candidate in (14e) instantiates the two schemas in (14a) and (14b). The expression given in (14f) is categorised by (14a) and (14d). Note that the schema in (14d) gives the context of reduction, i.e. the candidate (14f), which meets the context, conceptually overlaps with the schema to a high degree. As a consequence, its cognitive distance to the schema in (14d) is relatively close. Thus, the candidate in (14f) obtains additional activation value. Moreover, both candidates, (14e) and (14f), also receive activation value from the close relationship of the reduction schemas to the syllable schema. Nevertheless, the total activation value of (14f) is higher than

that of (14e) due to the more specific reduction schema it instantiates (cf. conceptual overlap). The final candidate in (14g) receives activation value from the schemas in (14a) and (14c). However, the reduction schema in (14c) is less specific than any of the other schemas. In other words, it exhibits a greater cognitive distance to the candidate and hence a lower degree of conceptual overlap. Therefore, the total activation value of (14g) is the lowest of all three. Since the *uniqueness* principle (1c) states that only the expression exhibiting the highest activation value is selected, (14e) and (14g) are deactivated and (14f) is correctly predicted as the winner. What remains to be discussed with respect to short vowels is reduction in syllables closed by sonorants. This will be considered in the following section.

# 7.1.2.3. Reduction in syllables closed by a sonorant

The patterns of vowel reduction in syllables closed by sonorants differs from the generalisations made above. Since reduction in such syllables is optional, a number of exceptions can be identified in the data set. However, exceptions to any of the generalisations established do not pose a problem to the theory as such and will be discussed in more detail in section 7.4. In order to cover for vowel reduction in word pairs such as *consult* /kənsált/ – *consultation* /kónsəltéjfən/ *consultation*, the analysis presented thus far needs to be expanded by a new schema. This is exemplified in the diagram in (15), which presents a sample analysis of the word /kónsəltéjfən/:

(15) Reduction in syllables closed by sonorants



The analysis presented in (15) shows four schemas in (15a-d). The first second-order schema in (15a) is a syllable schema specifying that sound strings such as (C)VC may be syllabified as closed syllables (cf. section 7.1.1). The schemas in (15b) and (15c) are two reduction schemas. While the former captures reduction to [ə], the latter states that a full vowel may retain its quality when stress is lost. The schema in (15d) ensures the reduction of mid-low vowels in syllables closed by sonorants. It is more specific than (15b) since it specifies the context to which it applies. Therefore, (15d) is compatible with and, in fact, even elaborates the higher-level schema (15b). As a result, it may also be referred to as a subschema of (15b) (cf. Nesset 2006; cf. section 7.3). Additionally, the box around (15a) and (15c) indicates that short vowels typically do not reduce in closed syllables.

Two candidate expressions, viz. [kónsəltéjʃən] and [kónsʌltéjʃən], are given in (15e) and (15f) respectively. The first candidate instantiates the syllable schema (15a), the reduction schema in (15b) and the reduction schema for closed syllables in (15d). By contrast, the second expression only instantiates two schemas, i.e. the closed syllable schema

(15a) and the reduction schema (15c). The competition between the two candidates is resolved in the same way as with the previous examples. Both candidate expressions are sanctioned by the first well-formedness principle (1a) access. The second principle, activation, allows for calculating the activation value of the candidates. The diagram clearly shows that the total activation value of (15e) is higher than that of the competing candidate (15f). On a simple schema count, the left candidate instantiates more schemas that the right expression. Furthermore, both schemas in (15b) and (15d) are more specific than (15c) and thus conceptually overlap with the candidate in (15e) to a higher degree. This, as was already mentioned, correlates with the close cognitive distance between the candidate (15e) and its categorising units. In contrast, the cognitive distance between (15f) to its schema (15c) is greater. What needs to be clarified at this point is the activation value added by the close connection between closed syllables and non-reduction indicated by the bold box. While the candidate in (15f) obtains additional activation value from this relation, it does not compensate for the lack of activation value resulting from the low degree of conceptual overlap. The schema in (15d) is a low-level schema, which, as a result of its high specificity, outweighs the effects of the close relationship.<sup>33</sup> It follows that (15e) obtains a higher amount of activation values from its categorising units and is correctly selected as the winning expression in the competition.

It remains without saying that the rather general analysis presented here for syllables closed by sonorants also applies to the vowel  $\epsilon$ / in word-initial syllables. In principle, the resulting analysis looks similar to the diagrams that have been discussed in this and the previous subsection and will thus not be considered in any more detail at this point. The additional schema that is needed to account for such cases is given in Figure 7d above. The model developed so far is capable of accounting for these word pairs in the data set as well. Another aspect worth emphasising is that while presented in isolation, the schemas in the diagrams should not to be understood as existing in a vacuum. As already mentioned, for reasons of simplicity, only the schemas relevant to the particular examples are represented in the diagrams. However, it should be emphasised that CG assumes that schemas form vast networks in which all of the schemas discussed interact with each other. Thus, even though

<sup>&</sup>lt;sup>33</sup> Remember that low-level schemas are given special importance in the approach, since they are directly abstracted from utterances. Thus, they can be assumed to contribute to a higher extent in the selection of the winning candidate.

separately introduced in the previous chapters, they should more aptly be represented together to emphasise the network-like structure they form. Having considered the reduction of short vowels in SSB, the thesis now moves on to the discussion of long monophthongs.

### 7.2. Reduction of long monophthongs

# 7.2.1. The schemas

Unlike short vowels, long monophthongs in unstressed position are indifferent to the type of syllable they occur in. Rather, reduction is sensitive to foot structure, viz. whether the unstressed syllable is footed or not (cf. section 6.2.2). Figure 8 below introduces two second-order schemas capturing foot structure in English. Since first-order schemas have been discussed in great detail in the previous section, they will be omitted from now on:



Figure 8 Foot-structure in English

The second-order schemas given in Figure 8 capture the generalisations speakers can make over sequences of stressed and unstressed syllables and possible foot structures. Put differently, speakers may form first-order schemas over sequences of stressed and unstressed syllables. This is shown in the upper boxes of Figure 8ab. They may also establish abstract schemas for how those syllables a grouped into feet, which is given in the lower boxes. Since English shows a systematic relationship between sequences of syllables and foot structure, language users may form second-order schemas over such relations. It should be emphasised that, as with syllable structure, foot structure in English is more complex than the relatively simple statements made by the schemas in Figure 8 above. However, the proposed schemas do not lay claim to giving a comprehensive account of footing in English. Rather, they are used to account for vowel reduction in SSB.

To capture the reduction of long monophthongs accurately, the reduction schemas suggested in section 7.1.1 need to be modified slightly. Due to their longer duration, long monophthongs (and diphthongs; cf. section 7.3) are schematically represented as VV. Figure

9 introduces the two different reduction schemas needed in the analysis presented below. First-order schemas are again deliberately omitted:



Figure 9 Reduction schemas for long vowels

As can be seen in Figure 9, two different schemas are necessary to account for the reduction behaviour of long monophthongs. The schema in Figure 9a generalises over all long vowels reducing to [ə] in SSB. However, since long monophthongs can also retain their full quality in unstressed position, an additional generalisation, viz. the schema in Figure 9b, is necessary. Note that second-order schemas are established on the basis of first-order generalisations over actual utterances and that no context is given as to where reduction occurs. The context of reduction, viz. footing, is taken care of by the foot structure schemas proposed in Figure 8 above. Having established the schemas needed, the thesis will now turn to the analysis.

### 7.2.2. Analysis

The diagram in (16) below presents an analysis of the reduction of long monophthongs in SSB. Since foot structure and reduction crucially dependent on each other, the foot structure schema in Figure 8a and the two reduction schemas in Figure 9 are necessary. The diagram in (16) presents an analysis for the word pair *conserve* /kənsə́:v/ – *conservation* /kónsəvéjʃən/:

(16) Reduction of long monophthongs



The diagram in (16) contains three second-order schemas. The first well-formedness principle, i.e. *access* (cf. section 3.2.2), sanctions the actualisation of two candidate expressions, viz. (16d) and (16e). As was done in the analysis of short vowel reduction, a bold box is added to indicate closely related schemas. In other words, the bold box around the foot schema in (16a) and the reduction schema in (16b) graphically represents the fact that long monophthongs typically reduce when they occur inside a phonological foot. Moreover, dashed lines range from the unstressed syllable in the lower part of (16a) to the lower boxes of (16b) and (16c). This is needed to spell out the relation between those schemas more directly, thus specifying the position of the unstressed syllable in the foot.

The calculation of the respective activation values of each candidate in (16) is relatively straightforward. Consider the candidate in (16d) first. Since it is categorised by two schemas, i.e. (16a) and (16b), it receives activation value from each. Moreover, the relatively strong relation between (16a) and (16b) further increases the activation value of the candidate expression. The candidate in (16e) also instantiates two schemas, viz. (16a) and (16c). However, these are not in any special relationship to each other. Thus, (16e) exhibits a relatively low total activation value. Consequently, the well-formedness principles in (1) correctly select the candidate in (16d) as the winner of the competition. It should be mentioned that the proposed analysis for long monophthongs in principle does not differ from the analysis of short vowels discussed in section 7.1. Rather, all that is needed to account for the difference observed in the reduction patterns is a different set of schemas generalising over the relevant factors. Moreover, this also demonstrates one of the advantages of CG, namely that no ad-hoc mechanisms are needed to propose a principled theory of vowel reduction.

What is left to be accounted for with respect to long monophthongs is the retention of full vowel quality in syllables not integrated into the foot-structure of a word. In order to capture this fact of the phonology of SSB, another foot schema, viz. the schema given in Figure 8b, is needed. The diagram in (17) below exemplifies how this issue can be dealt with in CG:

(17) Non-reduction of unfooted syllables



In principle, the analysis proposed in (17) does not differ from (16) above. The only difference can be found in the foot schema in (17a) and the bold box surrounding it and the reduction schema in (17c). The dashed lines connect the unstressed syllables and specify their position in the foot. In terms of activation value, the calculation largely follows the calculation discussed for the diagram in (16). The two schemas in (17a) and (17c) and the box indicating the strong relation between quality retention and the unfootedness of a syllable add activation value to the candidate in (17e). Since the total activation value obtained by the expression given in (17d) is lower than that of (17e), it is consequently deactivated. Thus, the model predicts the correct winner (17e), which is given in bold. Note that in the diagrams in (16) and (17), the reduction schemas are equally specific. Therefore, the decisive factor in the reduction or non-reduction of long monophthongs lies in the interaction of the second-order schemas, viz. the strong relation between foot structure and vowel reduction. The following section will now turn to the analysis of diphthongs.

## 7.3. Reduction of diphthongs

#### 7.3.1. The schemas

Only few straightforward generalisations such as those presented in the previous sections can be made for the reduction of diphthongs in SSB. In principle, foot structure is one of the conditioning factors necessary to account for the phenomenon. Thus, the foot schemas established in Figure 8 in section 7.2.1 also apply to diphthongs and will not be discussed in this section anymore. However, the reduction processes that can be observed for diphthongs are extremely complicated. While two reduction schemas have already been discussed with respect to the reduction of long monophthongs, viz. Figure 9, a number of additional lower-level, i.e. more specific, schemas are required to cover the facts discussed in section 6.2.3. These are given in Figure 10 below. Note, again, that no first-order schemas are given:<sup>34</sup>



Figure 10 Reduction schemas for diphthongs

In total, Figure 10 shows seven different lower-level schemas. Moreover, all of these schemas are relatively specific, as they determine the kind of vowel they refer to and, in some cases, also the context. While not economical, proposing such schemas is in agreement with the maximalist nature of CG and the importance that low-level schemas are given in the approach. Consider the schemas in Figure 10ab first. They capture the fact that /tj/ reduces to either [1] (in word-initial position) or [ə] (elsewhere). The two schemas in Figure 10c and 10d capture the alternation between [ə] and [1] for the diphthong /aj/, i.e. that [1] occurs as the elsewhere case as opposed to [ə], which is only found in sandhi contexts. The schema in Figure 10e generalises over the reduction of the vowel / $\varepsilon$ j/ to [ə] in all environments when stress is lost. The final two schemas in Figure 10f and 10g are needed to prevent the diphthongs /oj/ and /aw/ from reducing. The thesis will now turn to the analysis of diphthongs.

<sup>&</sup>lt;sup>34</sup> In principle, a global schema  $VV \rightarrow [I]$  may be added to Figure 10. However, since it is not necessary to predict the correct outcome of each competition, it is not used in the analyses. The relatively specific low-level schemas given suffice in the analyses to follow.

### 7.3.2. Analysis

Since the reduction patterns of diphthongs are relatively complex, not all relevant diagrams can be given in this section. However, each type of reduction, viz. reduction to [ə], reduction to [1], [ə]/[1] alternation and the non-reduction of /oj/ and /aw/ will be discussed in turns. An analysis of the vowels /uw/ and /əw/ will not be presented in this section. In principle, the behaviour of /uw/ and /əw/ in unstressed position does not differ from the reduction patters analysed with respect to long monophthongs. Consequently, graphical representations would be identical and only restate the aforementioned analysis, but not yield any more insight into the phenomenon. The following section will focus on the reduction of the most straightforward diphthong, viz. /ɛj/.

# 7.3.2.1. The diphthong /ɛj/

A diagram exemplifying the reduction of  $\epsilon_j$  for the word pair *able*  $\epsilon_j$  bal/ – *ability*  $\epsilon_j$  bilatij/ is given in (18) below. Note that no foot schemas are included, since  $\epsilon_j$  reduces in unstressed syllables regardless of whether the syllable is footed or not:

(18) Reduction of /ɛj/



The diagram in (18) is comparably simple. It contains three schemas, two of which are rather abstract reduction schemas, i.e. (18a) and (18b). Moreover, a more specific schema is given in (18c). (18c) is a type of schema that Nesset (2006) refers to as a *subschema*. To clarify

what subschemas are, it should be emphasised again that CG is a bottom-up approach. Speakers of a particular language form low-level schemas, i.e. schemas that are specific and generalise over a small set of utterances. On basis of such "local schemas" (2006: 59), more abstract schemas capturing a wider range of data, also called "global schemas" (2006: 59), are established. Consequently, the schema in (18c) is a local subschema from which the schema in (18a) is abstracted.<sup>35</sup>

The competition between the two candidate expressions given in (18d) and (18e) is resolved by the well-formedness principles. The reduction schemas in (18a) and (18b) may be considered global schemas. Neither of them specifies the context in which reduction takes place. Calculating the activation value of each candidate is simple. The expression in (18d) instantiates both the global schema in (18a) and the local subschema in (18c). Consequently, this candidate conceptually overlaps with its categorising unit to a high degree, i.e. the cognitive distance is rather close. By contrast, since the cognitive distance between the expression in (18e) and its categorising unit (18b) is relatively long, the total activation value of (18e) can be considered low. It follows that the model correctly predicts the winning candidate (18e), which is given in bold. The following subsection will turn to the reduction of the diphthong /aj/.

# 7.3.2.2. The diphthong /aj/

The reduction patterns of /aj/ in unstressed positions are more complex than those of / $\epsilon$ j/. The diagram in (19) below exemplifies the reduction of /aj/ to [I] on the word pair *horizon* / hərájzən/–*horizontal* /hśrizóntəl/:

<sup>&</sup>lt;sup>35</sup> In principle, subschemas could have been used in any of the analyses presented before. However, they have been omitted as they are, strictly speaking, not necessary to determine the winning candidate in the aforementioned analyses.

 $(19)/aj/\rightarrow [I]$ 



Since the reduction of /qj/ depends on whether it is integrated into the foot structure of a word or not, the graphical representation in (19) contains three schemas, i.e. a foot structure schema in (19a) and two reduction schemas in (19b) and (19c). Additionally, two candidate expressions (19d) and (19e) are given. Considering the well-formedness principles, it follows that the candidate in (19d) is selected as the winning expression. It not only instantiates the schemas in (19a) and (19b), but also conceptually overlaps with (19b) to a high degree. Thus, it obtains a higher amount of activation value than its competing expression (19e). In addition to reduction to [I] in context not specified any further, /qj/ also reduces to [ə] in the environment of r-sandhi in words such as /ədmájə/ – /ádməréjʃən/ (cf. section 6.2.3). A possible analysis of this pattern will not be presented separately at this point, as the only difference to the analysis presented above lies in the additional schema, i.e. Figure 10d. Figure 10d presents a subschema of the general reduction schema VV  $\rightarrow$  [ə], viz. /aj/  $\rightarrow$  [ə]. Since it not only states the outcome, but also the context of the reduction process (r-sandhi), it is more specific than the schema capturing reduction to [I]. Consequently, the activation value obtained by a potential candidate exhibiting [ə] outweighs the total activation value of the

wrongly reduced expression. The model thus also correctly predicts the reduction of /aj/ to [ə]. The subsequent section will turn to the analysis of the diphthong /ij/.

7.3.2.3. The diphthong /ıj/

The diagram in (20) below captures the reduction patterns for /ij/ in unstressed position on the example of *equal* /ijkwəl/ – *equality* /ikwólətij/:





The representation in (20) contains two different types of schemas. One the one hand, a footstructure schema, i.e. (20a) is given since the outcome of the reduction process relies on whether the vowel occurs in footed or unfooted position. On the other hand, three competing reduction schemas, i.e. (20b), (20c) and (20d) are found as well. They cover the different vowel qualities in unstressed syllables. In addition to these schemas, three candidate expressions, viz. (20e), (20f) and (20g) are actualised outside the grammar. The competition between the candidates is resolved in a relatively straightforward way. The candidate expression in (20g) obtains the lowest amount of activation value from its categorising unit. It only instantiates the relatively global reduction schema in (20d). What is more interesting to consider is the competition between the two candidates in (20e) and (20f). While the number of schemas categorising them is identical (including the strong relationship between reduction and foot structure indicated by the bold box), they considerably differ with respect to the degree of conceptual overlap. In other words, the schema in (20b) is more specific than the schema in (20c) (indicated by the graphical position in the diagram). Therefore, its candidate (20e) conceptually overlaps with it to a high degree, i.e. shows a close cognitive distance to its categorising unit. What follows is that it obtains a higher amount of activation value than any of the other candidates and is correctly selected as the winning candidate.

Although no diagram will be presented for the reduction to [ə], a number of comments on how it can be accounted for in CG are useful. It was established in section 6.2.3 that [ə] only occurs word-medially (or, put differently, as part of a foot). As its distribution is not further restricted (unlike that of [I], which only occurs in word-initial unstressed syllables), it may be considered the elsewhere case. In principle, the analysis of reduction to [ə] is identical to the analysis of long monophthongs presented in section 7.2. The schemas needed are the foot structure schema in Figure 8a and the two reduction schemas in Figure 9a and 9b. Consequently, competition only exists between a candidate showing a full vowel and a candidate exhibiting [ə]. A candidate showing reduction to [I] is logically not possible in this case, since it only applies to the phonotactic environment  $\#\sigma$  (which is not given in words reducing to [ə]). As vowel reduction and foot structure are closely related to each other, the reduced candidate wins the competition. Having discussed the prototypical cases of diphthong reduction, the thesis now moves on the non-reduction of /oj/ and /aw/.

# 7.3.2.4. The diphthongs /oj/ and /aw/

The diphthongs /oj/ and /aw/ retain their full quality irrespective of foot structure. How this can be dealt with in CG is exemplified in (21) below on the word pair *exploit* /ɪksplójt/ – *exploitation* /ɛ́ksplojtɛ́jʃən/:<sup>36</sup>

 $<sup>^{36}</sup>$  Since foot structure does not influence the reduction of /oj/ and /aw/, foot structure schemas have been excluded to make the representation simpler. It should be emphasised, however, that foot structure schemas may be added to the diagram to give a more comprehensive picture of the phenomenon.

(21) Non-reduction of /oj/



The analysis presented in (21) is essentially identical to the analysis given in (18) above. It contains two reduction schemas, i.e. (21a) and (21b), and one subschema, (20c). Moreover, two candidate expressions are given in (21d) and (21e). The calculation of the activation value is yet again relatively simple. The schemas in (21a) and (21b) are identical with respect to their respective degrees of specificity. Consequently, they cannot decide the competition. However, a more specific, local subschema (21c) is contained in the mental grammar as well, which elaborates the reduction schema in (21a) further. Consequently, the activation value obtained by the candidate in (21d) is higher. The expression conceptually overlaps with the more specific subschema to a higher degree, i.e. shows a closer cognitive distance to the categorising unit than its competing candidate and is thus selected as the winner of the competition. Thus far, regular cases of vowel reduction have been accounted for. It was shown that no ad-hoc mechanisms are necessary to propose a unified analysis of vowel reduction in SSB. What remains to be discussed is how frequency effects and semantic transparency can be incorporated into the theory developed here. In the following section, some exceptions will be considered in more detail.

# 7.4. Exceptions: Frequency effects in CG

While not all factors influencing the unexpected (non-)reduction of vowels can be discussed in this thesis, a possible analysis of frequency effects and semantic transparency within the framework developed here will be proposed. First, the non-reduction of low frequency items will be considered in section 7.4.1. Section 7.4.2 will then examine how high frequency items can be accounted for in CG.

### 7.4.1. Non-reduction of low frequency items

Frequency effects have a considerable effect on whether a given vowel reduces in unstressed position or not. The data set suggests that infrequent words tend to preserve the quality of vowels in contexts in which they would otherwise be expected to reduce. As discussed in section 2.2, CG assumes that each conventional linguistic unit, i.e. instantiation, is listed in the grammar of native speakers of a particular language. Conventional linguistic units have not been employed in the analysis so far to show that in principle, they are not needed to predict the correct winning candidate. However, in order to explain the non-reduction of particularly infrequent items (checked using CELEX), reference has to be made to units contained in the grammar. The diagram in (22) below illustrates how the non-reduction of low frequency items may be accounted for in the framework developed in this thesis on the example of the word-pair *passiv* /pásiv/– *passivity* /pasívətij/:

(22) Non-reduction of low frequency items



The diagram in (22) contains three schemas, viz. a syllable schema in (22a) and two reduction schemas in (22b) and (22c). Moreover, a conventional linguistic unit, i.e. /pásiv/, in

(22d) and two candidate expressions in (22e) and (22f) are given as well. The bold box around (22a) and (22b) indicates that short vowels in open syllables typically reduce in English. Note that the categorisation relationship between the unit in (22d) and the candidate in (22f) is of the type extension, which is indicated by the dashed arrow in bold. Put differently, they do not fully instantiate each other, but rather are only compatible to a certain degree (since there is a change in the stress pattern due to affixation).

The question raised by the diagram in (22) is how the competition between the two candidate expressions is resolved to give the correct prediction. On the one hand, the candidate in (22e) obtains activation value from both the syllable schema in (22a) and the reduction schema in (22b). Furthermore, it should be noted that that the relationship between those two schemas is relatively strong, since vowel reduction typically occurs in open syllables. Additional activation value is added by the fact that the schema in (22b) is more specific than its competing reduction schema in (22c). On the other hand, the candidate expression in (22f) instantiates the syllable schema (22a) and the less specific schema in (22c). In addition, the conventional linguistic unit adds activation value to the candidate (22f). In the light of this, the calculation of the total activation value of each candidate proves to be somewhat more complicated, since the well-formedness principles on their own do not suffice. Rather, semantic transparency needs to be taken into account to solve the competition.

When comparing the candidates to the schemas in their mental grammar, speakers may refer to the base word *passive* for reference (indicated by the dotted bold arrow ranging from (22d) to (22f)), from which additional activation value is obtained. Consequently, the activation value needed for the selection of (22f) as the winner is obtained from the fact that the word pair *passive – passivity* is semantically highly transparent (note the dashed arrow in bold). Since the meaning of the derivative can be arrived at by the meaning of the base, the relation between the two words is highly present in the minds of speakers. In other words, the close interlexical relation to the base word *passive* may explain the retention of the full vowel in the first syllable of *passivity* (cf. Kumashiro & Kumashiro 2006 on interlexical relation and stress). The candidate in (22e) does not refer to any conventional linguistic expression and hence is reduced as expected. It should be noted that what has been discussed here on the example of the word-pair *passive – passivity* in principle applies to any word-pair with a
comparable frequency count. The following section will discuss vowel reduction in high frequency derivatives.

# 7.4.2. Reduction of high frequency items

The unexpected reduction of highly frequent words cannot be accounted for by assuming interlexical relations. Rather, the notion of entrenchment, which was briefly discussed in section 2.1.2 becomes pivotal in this respect. In principle, entrenchment can be thought of as an effect of frequency. In usage-based approaches, it is generally assumed that frequency positively correlates with the degree of entrenchment of a particular linguistic unit (cf. Dąbrowska 2004). Another factor possibly influencing vowel reduction of highly frequent words is semantic transparency. Many of the words that show reduction in contexts that usually do not allow for it may be considered relatively opaque in terms of their semantics. Since the meaning of semantically intransparent derivatives is hardly to be arrived at by simply looking at the base and the added affix, it may well be assumed that these items behave somewhat more independently from their base than highly transparent words. Put differently, speakers may not be consciously aware of the interlexical relation to the base, which may then increase the likelihood of reduction of unstressed syllables. Moreover, together with high frequency, this may then result in the tendency for reducing syllables in context that usually would block reduction.

The diagram in (23) below proposes a possible analysis of the word pair *particle* / pá:tikəl/ – *particular* /pətíkjələ/. Note that *particular* is not only highly frequent in use (CELEX), but semantically opaque in relation to its base *particle*:

#### (23) Reduction of high frequency items



A number of different schemas is shown in the diagram in (23). Except for the subschema in (23d), all of them have been discussed in section 7.2 on long monophthongs. The box around schemas (23a) and (23c) graphically expresses the tendency for unfooted syllables not to reduce. Moreover, a conventional linguistic unit, viz. /pətíkjələ/, is contained in the grammar as well. Since /pətíkjələ/ is a highly frequent and thus entrenched unit, it is represented in bold. Furthermore, two candidate expressions are given outside the grammar.

Calculating the activation value of each candidate is trivial with respect to (23). The highly entrenched linguistic unit given in (23e) directly licences the candidate in (23f) and selects it as the winning candidate. It is important to note that the competition can only be solved with reference to the unit (23e), since the activation value obtained by the

candidates from their respective schemas may be considered equal. To spell it out more precisely, the candidate in (23f) receives activation value from (23a), (23b) and (23d). Its competing expression, however, instantiates the schemas in (23a) and (23c). Additionally, it also obtains activation value from the large box around those schemas. It may be true that the schemas in (23b) and (23d) are more specific and hence contribute to the total activation value to a larger extent that the remaining two schemas. Nevertheless, such an analysis would then not take into account the high frequency of the item *particular* and thus fail to account for frequency effects in the framework. Frequency is a property of the item itself and therefore cannot be accounted for without referring to the word in question. It is crucial to understand that only exceptions based on high frequency are explained that way. None of the analyses presented in the previous sections depends on linguistic units listed in the grammar. Rather, the selection of the winning candidate is entirely based on categorising relationships and conceptual overlap.

At this point, it should be mentioned that it is not possible to discuss all the relevant exceptions as part of this thesis. The diagrams presented in this section only present two cases in which interlexical relations, i.e. the relationships between individual linguistic units, and frequency effects can be said to influence atypical reduction patterns. In principle, however, all items in the data set showing similar patterns, viz. high or low frequency, may be analysed in much the same way. If a particular derivative is infrequent but semantically transparent with respect to its base, it may be assumed that speakers refer to the base when retrieving the unit. By contrast, a highly frequent and semantically opaque derivative is less strongly related to its source and may consequently be more likely do undergo reduction. It remains without saying that counterexamples to the analyses of exceptions presented here do exist. As was mentioned in the introduction to this thesis, phonological work within CG is still in its beginning phase. Consequently, many issues remain to be discussed in the literature. Nevertheless, it was shown so far that CG does possess the needed theoretical constructs to account for phonological phenomena. Thus far, only vowel reduction in English was examined. The next section will turn to an analysis of Russian and show that it can be straightforwardly handled by CG by the same theoretical constructs used in the previous sections on English.

## 8. Vowel reduction in Russian: a brief sketch

The aim of this section is to provide an analysis of Russian vowel reduction in the framework developed in this thesis. Moreover, it will be shown that the same theoretical constructs developed in the previous sections can successfully be applied to Russian as well. The structure of this section is as follows. Section 8.1 will briefly introduce the variety of Russian, viz. Contemporary Standard Russian (CSR), and discuss the type of data collected. Section 8.2 will examine those aspects of the phonology of Russian which are crucial for an understanding of vowel reduction. This will be followed by the presentation and discussion of the data in section 8.3. The final section of this part of the thesis proposes an analysis of vowel reduction in Russian couched in CG.

## 8.1. Contemporary Standard Russian

The language variety studied in what is to follow is commonly referred to as *Contemporary Standard Russian* (CSR) in the English-speaking literature (cf. Comrie, Stone & Polinsky 1979). By and large, the term CSR is used to refer to "the standardized language whose norms started to form in the late 1800s and stabilized by the middle of the twentieth century." (1979: 3). The data considered in this section was collected on the basis of the most contemporary pronunciation dictionary available for Russian, viz. *Bol'shoi orfoepicheskii slovar 'russkogo îazyka. Literaturnoe proiznoshenie i udarenie nachala XXI veka: norma i eë varianty* [Large pronunciation dictionary of the Russian language. Standard pronunciation and stress at the beginning of the 21<sup>st</sup> century: the norm and its variants] (Kalenchuk, Kasatkin & Kasatikina 2017). However, it should be noted that the dictionary does not give complete transcriptions of each entry. Rather, only segments deviating from the norm are explicitly transcribed. Therefore, the data used in this section was transcribed manually according to the rules of CSR.<sup>37</sup> The following section will now introduce the sound system of Russian.

#### 8.2. The Russian sound system

#### 8.2.1. Vowel system

Standard Russian has five different vowel phonemes in stressed position, viz. /a e i o u/ (cf. Jones & Wand 1969; Jaworski 2010; Iosad 2012 and others). A sixth vowel, i.e. [i], has been

<sup>&</sup>lt;sup>37</sup> This may be considered problematic in a usage-based approach. However, the focus of the present thesis is on vowel reduction in SSB. Russian is only included to show that no ad-hoc mechanisms are needed to account for the same phenomenon in a different language.

subject to longstanding discussions with respect to its status in the vowel system of Russian. While regarded as a distinct phoneme by some researchers working on Russian (e.g. Mołczanow 2008; Kasatkin 2003), others treat [i] as an allophone of the phoneme /i/ (e.g. Crosswhite 2001/2001; Padgett & Tabain 2005; Jaworski 2010; Iosad 2012). Figure 11 illustrates the vowel system found in Russian:



Figure 11 The vowel system in Russian (based on Akišina & Baranovskaya 1980: 86)

The vowel chart in Figure 11 shows the five vowels found in Russian. Since many scholars (e.g. Akišina & Baranovskaya 1980; Kasatkin 2003 and others) take [i] to be phonemic, it is included in the vowel chart in brackets. This, however, is not the approach followed here. The decision to treat [i] as an allophone of /i/ is based on the fact that both [i] and [i] are in complementary distribution: "[T]here is one phoneme *i*, which is realized as i after plain (non-palatalized) consonants, so long as no pause intervenes, that is, within something like the phonological phrase" [original emphasis] (Padgett 2001: 191). What follows from this definition is that [i] can never appear in word-initial syllables, unless /i/ is immediately preceded by a plain, i.e. non-palatalised, consonant. Two examples supporting this view may be given. First, *čital im* 'he read to them' is pronounced [tʃ1'tal im] and not \*[tʃ1'tal im] (Timberlake 2004: 40), as it is directly preceded by a plain /l/.<sup>38</sup> Another piece of evidence for the allophonic status of [i] is the alternation *Ivan* 'Ivan' [1'van] – *k Ivanu* 'to Ivan' [k i'vanʊ]. Many more examples of that sort can be found in Russian. Since the evidence for the allophonic status of [i] in Russian is relatively strong and counterexamples rare, [i] will be considered an allophone of /i/.

<sup>&</sup>lt;sup>38</sup> While Timberlake (2004: 40) does not explicitly mention it, it should be noted that *im* 'them' in isolation is pronounced [im]. Thus, a clear [i]/[i] alternation can be observed.

## 8.2.2. Consonant system

The inventory of consonants in Russian is relatively complex. A basic dichotomy can be identified between palatalised and non-palatalised consonants, which typically occur in pairs (Kasatkin 2003: 47). Table 7 below outlines the consonant system of CSR: **Table 7** The Russian consonant inventory (based on Akišina & Baranovskaya 1980: 57)

	paired/mutable	not paired/immutable
non-palatalised	p b v f t d m n l r s z k g x	$\int \mathfrak{Z} ts$
palatalised	pi bi vi fi ti di mi ni li ri si zi ki gi xi	t∫ j ∫: 3:

In Table 7, it can be seen that the majority of consonants in Russian are paired, i.e. they occur both in non-palatalised and palatalised forms. Seven consonants appear in one form only. It should be emphasised that palatalisation is regarded a feature which is intrinsic to consonants and not to vowels (Timberlake 2004: 57).<sup>39</sup> This assumption is based on the fact that palatalisation has a contrastive function in word-final position where no vowel follows, e.g. [getov] 'ready' vs. [getovj] 'prepare' and ['vip<sup>i</sup>tt] 'drunk down' vs. ['vip<sup>i</sup>tt<sup>j</sup>] 'to drink down' (2004: 57). Having discussed the basics of Russian phonology, the thesis will now move on to present the data considered in the analysis.

## 8.3. Vowel reduction in Russian: the data

Vowel reduction in Russian is determined by three factors, viz. "the identity of the underlying segment, its position within the word and the palatalization or lack thereof of the consonant preceding the vowel" (Iosad 2012: 522). No need for underlying segments arises, if word pairs consisting of a base word and a derivative (or inflected word form) in which stress changes are considered. Thus, the present thesis takes the identity of the vowel in the base word as one of the factors influencing the outcome of reduction in Russian. Out of the five stressed vowels, only /a o e/ reduce in Russian (cf. Iosad 2012). The vowels /i u/ do not undergo any phonological changes in unstressed syllables but are strongly centralised (Iosad 2012: 524). Moreover, Russian is traditionally considered a language exhibiting two degrees of qualitative reduction, which are termed *moderate* and *radical reduction* (Crosswhite 2000: 109). The degree to which a vowel is reduced depends on the distance of the unstressed

<sup>&</sup>lt;sup>39</sup> Other approaches (see, for example, Lighter 1972) consider consonants intrinsically hard. Palatalisation is consequently treated as an effect of front vowels on consonants. However, such approaches seem to be generally rejected nowadays and will not be considered any further.

vowel from the stressed syllable. Moderate reduction occurs in the first pretonic syllable and in any onsetless syllable regardless of its position (Iosad: 2012: 524). Radical reduction can most successfully be considered the elsewhere case. It occurs in syllables with an onset not directly preceding the stressed syllable (2012: 524). The last factor mentioned in the quote above is the palatalisation of the preceding consonant. Palatalisation further divides the reduction patterns into two subgroups, which are outlined in (24) below:

(24)	a.	Moderate Reduction	b.	Radical Reduction
		$C^{j}/a o e \rightarrow [I]$		$C^{j}/a \ o \ e/ \rightarrow [I]$
		$C_a o \to [v]$		$C_/a o \to [a]$

As can be seen in 24, /a o e/ reduce to [I] after palatalised consonants irrespective of their position. After non-palatalised consonants, only /a o/ reduce; to [v] in moderate and to [ə] in radical reduction (cf. Iosad 2012; Padgett & Tabain 2005).<sup>40</sup> Consequently, Russian exhibits both centripetal, i.e. contrast-reducing, and centrifugal, i.e. contrast-enhancing reduction, patterns.

The word pairs in Table 7 below illustrate vowel reduction in Russian in both contexts, viz.  $C^{j}$  and  $C_{f}$  for all three vowels qualified for reduction. Since the aim of this section is simply to demonstrate the potential of the approach outlined in the previous section, a comprehensive data set was not collected. Rather, the word pairs given in Table 8 below may be considered representative for the reduction processes in Russian: Table 8 Vowel reduction in Russian

Moderate reduction C <sup>j</sup> _		Radical Reduction C <sup>j</sup> _		
'p <sup>j</sup> atj	'five'	'p <sup>j</sup> at <sup>j</sup>	'five'	
p <sup>j</sup> ı'tji	'five (gen.sg.)'	p <sup>j</sup> ıtɐ't∫ok	'five-kopeck piece'	
'l <sup>j</sup> es	'forest'	'd <sup>j</sup> es <sup>j</sup> ıt <sup>j</sup>	'ten'	
l <sup>j</sup> 1'sa	'forest (gen.sg.)'	d <sup>j</sup> ıs <sup>j</sup> ı't <sup>j</sup> i	'ten (gen.sg.)	
'n <sup>j</sup> os n <sup>j</sup> ıs'la	'he carried' 'she carried'			

<sup>&</sup>lt;sup>40</sup> A note on Russian stress placement is required at this point. Stress is Russian is considered free and not predictable on basis of any rules (cf. Thelin 1971). In principle, stress can be placed on any syllable and any morpheme in a word, i.e. it can fall on prefixes, roots, suffixes or endings (Kasatkin 2003: 67). Hence, it is assumed that "Russian morphemes are stored in the lexicon along with the corresponding information about their 'accented' (i.e. stressed) or 'unaccented' (i.e. unstressed) status" (Jouravlev & Lupker 2015: 945).

Moderate reduction C_	Radical reduction C_	
<ul><li>'kot 'male cat'</li><li>kɛ'ta 'male cat (gen.sg.)'</li></ul>	stɐˈrʲik stərʲɪˈka	ʻold manʻ ʻold man (gen.sg.)ʻ
ˈdavnʲɪj 'old' dɐv'no 'long ago (adv.)'	'gorət gəre'dok	<pre>`city` `small city`</pre>

Having briefly presented the main facts of Russian vowel reduction, the thesis will now show how these facts can be accommodated in a CG framework.

#### 8.3.4. Analysis of Russian vowel reduction

An analysis of vowel reduction in Russian from the perspective of CG can already be found in Nesset (2006). However, the analysis presented in this thesis differs in two ways. The first difference concerns the use of moraic theory. In order to explain the reduction patterns in Russian, Nesset (2006) uses the concept of the mora, but gives no arguments for the benefits of such an analysis. Rather, while morae are included in the schematic representations, their use remains uncommented. Since vowel reduction in Russian neither depends on syllable weight nor on vowel length (in fact, long vowels do not exist in Russian at all), the analysis proposed in this section does not refer to the morae. The second difference relates to the facts of reduction as presented by Nesset (2006). As was discussed in section 8.2 above, the vowels /i/ and /u/ are assumed not to reduce. Nesset (2006), however, takes a different position and argues for the reduction of /i/. In the light of more recent studies on the vowel system of Russian (e.g. Iosad 2012), the approach taken here acknowledges the strong centralisation of /i/ and /u/ in unstressed position, but does not propose the reduction of /i/. In the next section, the schemas underlying the subsequent analysis will be presented.

# 8.3.5. The schemas

The reduction patterns found in Russian are less complex than those for English. Reduction neither depends on syllable structure nor on foot structure. Hence, the schemas proposed in this section are comparably simple. Figure 12 below introduces the schemas used in the subsequent analysis. As first-order schemas have been discussed in detail with respect to reduction in English, only the second-order schemas are given:



Figure 12 Reduction schemas for Russian

Figure 12 above captures the facts of vowel reduction presented in section 8.3 in four different second-order schemas. The schema in Figure 12a covers the fact that high vowels in Russian, viz. /i/ and /u/, do not reduce. The remaining three schemas in Figure 12b-d cover mid-low vowels. Figure 12b and 12c generalise over moderate and radical reduction in non-palatalised contexts. Note that the schema in Figure 12c is necessarily more specific as it specifies the position of the stressed syllable as immediately following the reduced syllable. <sup>41</sup>The final second-order schema Figure 12d generalises over moderate and radical reduction in palatalised contexts. Since the outcome of reduction is identical irrespective of the distance to the stressed syllable, only one schema is needed to accommodate the phenomenon. It should be noted, however, that the reduction patterns in Russian are more intricate than can be discussed in this thesis. Only the prototypical patterns are covered for by the schemas in Figure 12. The following section will discuss how Russian vowel reduction can be explained in a CG framework.

## 8.3.6. Analysis

8.3.6.1. Moderate and radical reduction in non-palatalised contexts

Moderate reduction, i.e. the reduction of vowels in pretonic position, in contexts where no palatalised consonants are found can be accounted for by means of the diagram in (25) below. Since neither foot-structure nor syllable structure are relevant, the diagram is relatively simple:

<sup>&</sup>lt;sup>41</sup> Note that this necessarily also includes the vowel /e/, which does actually not reduce in non-palatalised contex

(25) Moderate reduction in C\_



As can be seen in (25), only two schemas are needed in the analysis of moderate reduction. The schema in (25a) specifies that mid-low vowels, i.e. /a o e/, reduce to [ə] when stress is lost. By contrast, the second-order schema in (25b) states the reduction of mid-low vowels to [v] if the tonic syllable immediately follows. Two candidate expressions are actualised in (25c) and (25d). Note that they differ with respect to the reduced vowel, viz. [ə] in (25c) and [v] in (25d). Moreover, it should be emphasised that the two schemas in the grammar are not equally specific. Rather, (25a) represents what is often referred to as the elsewhere case. Since there is no context that would restrict the application of the schema, it applies whenever there is no other more specific schema present. (25b), however, states the context and is thus considered more specific.

The calculation of the activation value of each candidate crucially depends on the recognition of the elsewhere case. Each of the candidates instantiates one schema in the mental grammar. However, they do not do so equally well. While (25c) instantiates the elsewhere case, the expression in (25d) is categorised by the more specific schema. Consequently, the latter conceptually overlaps with its schema to a higher degree. As a result, the cognitive distance between (25d) and its categorising unit is considered relatively close. By contrast, the candidate in (25c) instantiates the elsewhere schema in (25a). What follows from this is that the activation value of (25d) is necessarily higher than that of (25c) and thus is correctly selected as the winning candidate. An important point of the analysis presented

here is that it also makes predictions as to radical reduction in non-palatalised contexts. As (25b) specifies the context, it does not sanction the actualisation of a candidate expression not immediately followed by a stressed syllable. Thus, with respect to radical reduction, no competition emerges in the first place. Rather, as there is no more specific schema in the grammar that could give rise to a candidate, the candidate instantiating the elsewhere schema is automatically chosen as the winning expression.

One issue not yet resolved is moderate reduction in onsetless syllables. In order to cover for this, the schemas given in Figure 12 require some modification. For instance, an additional schema specifying that [v] is the outcome of reduction in such environments may be added to Figure 12. Moreover, the schema in Figure 12b may then be specified further to only cover syllables in which the onset slot is filled. There are other possibilities for capturing this aspect of vowel reduction in Russian as well. However, the data set collected for this part of the thesis does not yield conclusive evidence for this phenomenon. Therefore, it is not included in the analysis. Moreover, since the aim of this section is not to provide a detailed analysis, but rather to show that the framework developed in this thesis can straightforwardly explain phenomena in other languages, not all relevant facts were taken into the analysis. The next section will move on to reduction in palatalised contexts.

8.3.6.2. Moderate and radical reduction in palatalised contexts

Mid-low vowels directly preceded by a palatalised consonant reduce to [I]. An analysis of moderate reduction in palatalised contexts is given in (26) below:

(26) Moderate reduction in C<sup>j</sup>\_



The diagram in (26) contains three schemas, each of which sanctions the actualisation of one candidate expression. The schema in (26a) covers the elsewhere case mentioned in the previous section. Consequently, the candidate expression in (26d) shows the reduced vowel [ə]. (26b) gives the schema for moderate reduction after non-palatalised consonants. It categorises a candidate expression (26e) in which the vowel in question is reduced to [v]. The final schema in (26c) captures reduction after palatalised consonants. Thus, the candidate in (26f) contains [1] in unstressed position. Note that the schemas also differ with respect to their position in the mental grammar, i.e. they range from graphically furthest to graphically closest to the candidate expression. In other words, while the schema in (26a), i.e. the elsewhere schema, is the most general one and therefore exhibiting the highest distance from its candidate, the schema in (26c) can be considered most specific. Seeing that both (26b) and (26c) state the context of the reduced vowel in the lower box of the second-order schema, it may be asked as to why (26c) shows a higher degree of specificity. The answer to that question is found in the top box of (26c), which identifies the context needed in the base word. Put differently, while the upper box in (26b) applies to any unit with a mid-low vowel, the upper first-order schema in (26c) is more restrictive in that it allows for mid-low vowels in palatalised contexts only.

Calculating the activation value of each candidate expression requires Langacker's (1999) notion of conceptual overlap. As was mentioned, the schemas in (26) increase in specificity from left to right. Hence, (26d) conceptually overlaps with its schema to a relatively low degree, while (26f) shows a comparably high degree of overlap. What follows from this is that the expression in (26f) necessarily obtains the highest amount of activation value, i.e. the cognitive distance between the candidate and the schema is relatively close, and thus (26f) is correctly selected as the winning candidate in the competition. In addition to covering for moderate reduction in CSR, the diagram presented in (26) has implications for radical reduction in palatalised contexts as well. Since the outcome of moderate and radical reduction after palatalised consonants is identical, the schema in (26c) does not specify the position of the following stressed syllable. Consequently, it applies to both types of reduction. Thus, a diagram for radical reduction will not be given at this point. It suffices to mention that the elsewhere schema in (26a) and the palatalised reduction schema in (26c) stand in

competition with respect to radical reduction.<sup>42</sup> Conceptual overlap solves the arising conflict between the candidates in favour of the expression showing reduction to [1].

## 9. Conclusion

The aim of the present thesis was to propose a non-reductive analysis of vowel reduction in both Standard Southern British (SSB) and Contemporary Standard Russian (CSR) couched in the framework of cognitive grammar. Vowel reduction is understood as the neutralisation of "two or more [...] vowel qualities [...] in a stress-dependent fashion" (Crosswhite 2001: 3). Since underlying representations are prohibited by the content requirement, word pairs consisting of a base and a derivative with differing stress patterns were collected in both SSB and CSR. Thus, reduction processes could be observed by comparing the respective syllables to each other. The data set collected suggests that vowel reduction in English depends on syllable structure (short vowels) and foot structure (long monophthongs and diphthongs). While short vowels generally reduce in open syllables, closed syllables seem to block vowel reduction in unstressed positions. Long monophthongs and diphthongs undergo reduction when integrated into the food structure of the derivative. Consequently, vowels in wordinitial unstressed syllables tend to retain their full quality. Both centripetal and centrifugal patterns, i.e. reduction to [a] and to [1], were identified. Particularly for diphthongs, many counterexamples to the foot-based generalisations can be found. The thesis suggests that frequency effects and semantic transparency may explain the unexpected behaviour of vowels in unstressed syllables. The Russian data allow for more straightforward generalisations. Generally, Russian exhibits two degrees of reduction, viz. moderate and radical reduction (Crosswhite 2000: 109). Following a palatalised consonant, mid-low vowels in Russian reduce to [1] in both moderate and radical reduction. In non-palatalised contexts |a| and |o| reduce to [v] in pretonic position and to [a] elsewhere.

Having collected the respective data set, it was possible to establish generalisations and translate these into cognitive schemas. The competition between schemas and their candidate expressions was solved by a set of four well-formedness principles, calculating the total activation value of each expression. The candidate showing the highest amount of activation is selected as the winner of the competition. Moreover, it was shown that the same

<sup>&</sup>lt;sup>42</sup> As the schema in (26b) states that a stressed syllable immediately follows the reduced vowel, it does not give rise to any candidate expression in this case.

theoretical constructs, viz. categorising relationships and cognitive schemas, are able to account for reduction phenomena in two distinct languages without employing any ad-hoc mechanisms. Another objective of the present thesis was to make a case for including phonology in cognitive linguistic approaches. Phonological work has long been ignored within cognitive linguistics at large and cognitive grammar in particular. This exclusion of phonology from the realm of CG is mostly due to its long-standing emphasis on semantics. However, phonological phenomena are slowly beginning to be explored from a cognitive linguistic perspective (e.g. Kumashiro 2000; Kumashiro & Kumashiro 2006; Nesset 2006 and 2008 and others). Consequently, the present thesis also contributes to the rather scarce literature in the field.

The phonological framework proposed in the present thesis heavily relies on two cognitive processes, viz. schema formation and categorisation relationships. Schemas take in a prominent position in the analysis presented in this thesis. While speakers form first-order schemas over utterances they experience, they may also connect those first-order schemas and thus establish second-order schemas. Second-order schemas are indispensable in a CG account of phonological phenomena (cf. Nesset 2008). They capture so-called sourceoriented generalisations, i.e. they relate the outcome of a process to its source (e.g. stressed and unstressed syllables in related words). Moreover, it was repeatedly emphasised that schemas do not exist in a vacuum in the mental grammar of speakers. Rather, they form complex networks, in which they interact with, but also contradict each other. Schemas are related to other schemas by categorisation relationships. In principle, there are two different types, viz. instantiations and extensions (Langacker 1987: 371). Instantiations refer to situations in which one schema showing a greater degree of specificity is connected to a more general schema. Put differently, both are compatible and the more specific schema elaborates the less specific schema further. By contrast, extension describes a relationship between two schemas which are only partly compatible. For instance, second-order schemas typically connect two first-order schemas via extension, since they are not completely compatible. Schema formation and categorisation relationships are generally known cognitive mechanism not limited to phonological phenomena or linguistic cognition in general. (cf. Janda 2015).

The analysis presented in this thesis is grounded in usage-based linguistics (cf. Bybee 2001). Therefore, it does not assume the existence of underlying representations, but only

allows structures which are directly observable in language use. In this sense, the analysis takes a maximalist, non-reductive and bottom-up approach. Instantiations, i.e. concrete linguistic items, are stored in the grammar and function as the basis on which abstract schemas are established bottom-up. While such an approach may not be economical, studies (e.g. Pierrehumbert 2001, Bybee 2006) have shown that the human mind has an immense capacity for lexical storage. Moreover, usage-based approaches emphasise the impact frequency has on the mental representation of language (Bybee 2001: 6-7). Grammar is not a static entity in the minds of speakers, but instead is constantly being reshaped by the experiences speakers have with language. For instance, high frequency strengthens the representations of items in the speakers' minds, which are then more easily activated (Dabrowska 2004: 213). Consequently, highly frequent items are strongly represented in language users' mental grammar and can thus be assumed to licence themselves. By contrast, low frequency items are harder to retrieve since their representation is comparably weak. It was shown that lexical relations and semantic transparency may explain why a particular vowel does not reduce in low frequency items. In retrieving an infrequent derivative, speakers may have to refer to the base word and do consequently not reduce the unstressed vowel to keep the derivative close to the base form.

While the theoretical constructs proposed by Kumashiro (2000) and Nesset (2006, 2008) provide valuable starting points for studies into the matter, issues remain that call for further research. For instance, vowel reduction in English is closely related to the phonological structure of the word, i.e. open or closed syllables for short vowels and foot structure for long monophthongs and diphthongs. Consequently, the thesis proposes that this close relationship has an influence on the total activation value of a candidate expression. However, it is not yet clear whether such relationships can be treated as a third type of schema next to first- and second-order schemas. More research into how second-order schemas interact is needed to clarify the theoretical status of such relationships. Moreover, the study presented in this thesis is purely theoretical. It would be interesting to see how the theoretical claims put forward in this thesis could be tested empirically. A lack of empirical hypothesis testing is characteristic of cognitive linguistics in general (Dąbrowska 2016: 483). Thus, developing means of empirical testing in cognitive linguistics would provide an extremely fruitful area for investigations. Furthermore, testing claims about schema

formation would yield empirical support for a more cognitively oriented phonology. Nevertheless, the present thesis has shown possible ways in which phonological phenomena can be accounted for in CG. While much remains to be explored, it has been shown that CG provides the needed theoretical constructs for a unified theory of phonology.

# 10. References

- Akišina, Alla; Baranovskaja, Svetlana. 1980. Russkaja fonetika [Russian phonetics]. Moskva: Russkij jazyk.
- Anderson, Stephen R. 1985. *Phonology in the twentieth century: theories of rules and theories of representations*. Chicago, IL: University of Chicago Press.
- Barnes, Jonathan. 2006. *Strength and weakness at the interface: positional neutralization in phonetics and phonology*. Berlin: Mouton de Gruyter.
- Beckman, Mary E. 1986. Stress and non-stress accent. Dordrecht: Foris Publications.
- Bell, Melanie J.; Schäfer, Martin. 2016. "Modelling semantic transparency". *Morphology* 26(2), 157-199.
- Blevins, Juliette. 2004. *Evolutionary phonology: the emergence of sound patterns*. Cambridge: Cambridge University Press.
- Burzio, Luigi. 1994. Principles of English stress. Cambridge: Cambridge University Press.
- Burzio, Luigi. 2007. "Phonology and phonetics of English stress and vowel reduction". *Language Sciences* 29(2), 154-176.
- Bybee, Joan L. 2001. Phonology and language use. Cambridge: Cambridge University Press.
- Bybee, Joan L. 2006. "From usage to grammar: the mind's response to repetition". *Language* 82(4), 711-733.
- Bybee, Joan L.; Slobin, Dan I. 1982. "Rules and schemas in the development and use of the English past tense". *Language* 58(2), 265-289.
- *CELEX database*. http://web.phonetik.uni-frankfurt.de/simplex.html (18 Dec. 2019). See Wouden 1988.
- Chomsky, Noam; Halle, Morris. 1991 [1968]. *The sound pattern of English*. New York, NY: Harper & Row.
- Clements, George N. 2009. "Does sonority have a phonetic basis?" In Raimy, Eric; Cairns, Charles E. (eds.). *Contemporary views on architectures and representation in phonology*, Cambridge, MA: MIT Press, 165-175.
- Cohn, Abigail C. 2011. "Features, segments, and the sources of phonological primitives". In Ridouane, Rachid; Clements, George N. (eds.). *Where do phonological features come from? Cognitive, physical and developmental bases of distinctive speech categories.* Amsterdam: John Benjamins, 15-41.
- Comrie, Bernhard; Stone, Gerald; Polinsky, Maria. 2003 [1978]. *The Russian language in the* 20<sup>th</sup> century. (2<sup>nd</sup> edition). Oxford: Clarendon Press & Oxford University Press.
- Crosswhite, Katherine. 2000. "Vowel reduction in Russian: a unified account of standard, dialectal, and 'dissimilative' patterns". University of Rochester Working Papers in the Language Sciences 1, 107-171.
- Crosswhite, Katherine. 2001. *Vowel reduction in optimality theory*. New York, NY: Routledge.
- Dąbrowska, Ewa. 2004. Language, mind and brain: some psychological and neurological constraints on theories of grammar. Edinburgh: Edinburgh University Press.
- Dąbrowska, Ewa. 2016. "Cognitive linguistics' seven deadly sins". *Cognitive Linguistics* 27(4), 479-491.
- Ellis, Alexander. 1869. On early English pronunciation: with especial reference to Shakspere and Chaucer, containing an investigation of the correspondence of writing with speech in England from the Anglosaxon period to the present day, preceded by a systematic notation of all spoken sounds by means of the ordinary printing types;

including a re-arrangement of Prof. F. J. Child's memoirs on the language of Chaucer and Gower, and reprints of the rare tracts by Salesbury on English, 1547, and Welch, 1567, and by Barcley on French, 1521. London: Asher & Co.

- Flemming, Edward; Johnson, Stephanie. 2007. "Rosa's roses: reduced vowels in American English". *Journal of the International Phonetic Association* 37(1), 83-96.
- Fox, Anthony. 2000. *Prosodic features and prosodic structure: the phonology of 'suprasegmentals'*. Oxford: Oxford University Press.
- Fudge, Erik. 1984. English word-stress. London: Allen & Unwin.
- Goad, Heather. 2012. "s C clusters are (almost always) coda-initial". *The Linguistic Review* 29(3), 335-373.
- Giegerich, Heinz J. 1999. *Lexical strata in English: morphological causes, phonological effects*. Cambridge: Cambridge University Press.
- Gimson, Alfred C. 1974 [1970]. *An introduction to the pronunciation of English*. (2<sup>nd</sup> edition). London: Edward Arnold.
- Goudbeek, Martijn; Smits, Roel; Cutler, Anne; Swingley, Daniel. 2017. "Auditory and phonetic category formation". In Cohen, Henri; Lefebvre, Claire (eds.). *Handbook of categorization in cognitive science*. Amsterdam: Elsevier, 687-708.
- Gussenhoven, Carlos. 2006. "Between stress and tone in Nubi word prosody". *Phonology* 23(2), 193-223.
- Halle, Morris; Kenstowicz, Michael. 1991. "The free element condition and cyclic versus noncyclic stress". *Linguistic Inquiry* 22(3), 457-501.
- Hammond, Michael. 1997. "Vowel quantity and syllabification in English". *Language* 38(1), 1-17.
- Hammond, Michael. 1999. *The phonology of English: a prosodic optimality-theoretic approach*. Oxford: Oxford University Press.
- Harnad, Stevan. 2017. "To cognize is to categorize: cognition is categorization." In Cohen, Henri (ed.). Handbook of categorization in cognitive science. Amsterdam: Elsevier, 21-54.
- Harris, John. 2005. "Vowel reduction as information loss." In Carr, Philip (eds.). *Headhood, elements, specification and constrastivity: phonological papers in honour of John Anderson.* Amsterdam: Benjamins, 119-132.
- Harris, John; Lindsey, Geoff. 2000. "Vowel patterns in mind and sound." In Burton-Roberts, Noel; Carr, Philip; Docherty, Gerard (eds.). *Phonological knowledge: conceptual and empirical issues*. Oxford: Oxford University Press, 185-205.
- Hulst, Harry van der. 2010. "Word accent: terms, typologies and theories". In Hulst, Harry van der; Goedemans, Rob; Zanten, Ellen van (eds.). *A survey of word accentual patterns in the languages of the world*. Berlin: Mouton de Gruyter, 3-53.
- Hyman, Larry. 1985. A theory of phonological weight. Dordrecht: Foris Publications.
- Iosad, Pavel. 2012. "Vowel reduction in Russian: no phonetics in phonology". *Linguistics* 48(3), 521-571.
- Jacobs, Haike. 1989. "Non-linear studies in the historical phonology of French". PhD Thesis. Katholiek Universiteit, Nijmegen.
- Janda, Laura A. 2015. "Cognitive linguistics in the year 2015". *Cognitive semantics* 1(1), 131 154.
- Jaworski, Sylwester. 2010. "Phonetic and phonological vowel reduction in Russian". *Poznań Studies in Contemporary Linguistics* 46(1), 51-68.

- Jones, Daniel; Ward, Dennis. 1969. *The phonetics of Russian*. Cambridge: Cambridge University Press.
- Jouravlev, Olessia; Lupker, Stephen. "Predicting stress patterns in an unpredictable stress language: the use of non-lexical sources of evidence for stress assignment in Russian". *Journal of Cognitive Psychology* 27(8), 944-966.
- Kager, René. 1999. Optimality Theory. Cambridge: Cambridge University Press.
- Kalenchuk, Maria L.; Kasatkin, Leonid L.; Kasatkina, Rozaliia F. 2017. Bol'shoi orfoepicheskii slovar'russkogo iazyka. Literaturnoe proiznoshenie i udarenie nachala XXI veka: norma i eë varianty [Large pronunciation dictionary of the Russian language. Standard pronunciation and stress at the beginning of the 21st century: norm and its variants]. (2nd edition). Moscow: Ast-Press Shkola.
- Kasatkin, Leonid. 2003. *Fonetika sovremennogo russkogo literaturnogo jazyka* [The phonetics of the modern Russian literary language]. Mosvka: Izd. Moskovskogo universiteta.
- Kiparsky, Paul. 1982. "From cyclic phonology to lexical phonology". In Hulst, Harry van der; Smith, Norval (eds.). *The structure of phonological representations. Vol. 1.* Dordrecht: Foris Publications, 131-176.
- Keizer, Evelien. 2015. *A Functional Discourse Grammar for English*. Oxford: Oxford University Press.
- Kumashiro, Fumiko. 2000. "Phonotactic interactions: a non-reductionist approach to phonology." PhD Thesis. Department of Linguistics, University of California.
- Kumashiro, Fumiko; Kumashiro, Toshiyuki. 2006. "Interlexical relations in English stress". *International Journal of English Studies* 6(2), 77-106.
- Langacker, Ronald W. 1987. *Foundations of cognitive grammar*. Stanford, CA: Stanford University Press.
- Langacker, Ronald W. 1991. *Concept, image, and symbol: the cognitive basis of grammar*. Berlin: Mouton de Gruyter.
- Langacker, Ronald W. 1999. Grammar and conceptualization. Berlin: Mouton de Gruyter.
- Langacker, Ronald W. 2000. "A dynamic, usage-based model". In Barlow, Michael; Kemmer, Suzanne (eds.) Usage based models of language. Stanford, CA: CSLI Publications, 1-63.
- Langacker, Ronald W. 2010. "How not to disagree: the emergence of structure from usage". In Boye, Kasper; Engberg-Pedersen, Elisabeth (eds.). *Language usage and language structure*. Berlin: Mouton de Gruyter, 107-143.
- Lightner, Theodore. 1972. *Problems in the theory of phonology*. Edmonton: Linguistic Research.
- Lindsey, Geoff. 2019. *English after RP: standard British pronunciation today*. Cham: Palgrave Macmillan.
- Lindsey, Geoff; Szigetvári, Péter. 2013. *Current British English searchable transcriptions*. http://cube.elte.hu (1 Dec. 2019).
- Marchand, Hans. 1969. *The categories and types of present-day English word-formation: a synchronic-diachronic approach*. (2<sup>nd</sup> edition). München: C.H. Beck.
- Martínez-Gil, Fernando. 2001. "Sonority as a primitive phonological feature: evidence from Spanish complex onset phonotactics". In Contreras, Heles; Herschensohn, Julia; Mallén, Enrique; Zagona, Karen (eds.). *Features and interfaces in Romance: essays in honor of Heles Contreras*. Amsterdam: John Benjamins, 203-222.

Mielke, Jeff. 2008. *The emergence of distinctive features*. Oxford: Oxford University Press. Mompeán-Gonzáles, José. 2004. "Category overlap and neutralization: the importance of

speakers' classification in phonology". Cognitive Linguistics 15(4), 429-469.

Mołczanow, Janina. 2008. The phonology of glides in Russian. München: LINCOM Europa.

Nathan, Geoffrey S. 1996. "Steps towards a cognitive phonology". In Hurch, Bernhard; Rhodes, Richard A. (eds.). *Natural phonology: the state of the art*. Berlin: Mouton de Gruyter, 107-120.

Nathan, Geoffrey S. 2015. "Phonology". In Dąbrowska, Ewa; Divjak, Dagmar (eds.). *Handbook of cognitive linguistics*. Berlin: Mouton de Gruyter, 253-273.

Nesset, Tore. 2006. "Second-order schemas and active subschemas: vowel reduction in cognitive grammar." *Poljarnyj Vestnik* 9, 52-61.

Nesset, Tore. 2008. *Abstract phonology in a concrete model: cognitive linguistics and the morphology-phonology interface*. Berlin: Mouton de Gruyter.

- *OED: The Oxford English dictionary*. 2019. https://www.oed.com (18 Dec. 2019). Oxford: Oxford University Press.
- Padgett, Jaye. 2001. "Contrast dispersion and Russian palatalization". In Hume, Elizabeth; Johnson, Keith (eds.). *The role of speech perception in phonology*. San Diego, CA: Academic Press, 187-218.
- Padgett, Jaye; Tabain, Marija. 2005. "Adaptive dispersion theory and phonological vowel reduction in Russian". *Phonetica* 62, 14-54.
- Pierrehumbert, Janet. 2001. "Exemplar dynamics: word frequency, lenition and contrast". In Hopper, Paul; Bybee, Joan (eds.). *Frequency and the emergence of linguistics structure*. Amsterdam: John Benjamins, 137-157.
- Plag, Ingo. 2003. Word-formation in English. Cambridge: Cambridge University Press.
- Prince, Alan S.; Smolensky, Paul. 2004. *Optimality theory: constraint interaction in generative grammar*. Maldan, MA: Blackwell.

Pulvermüller, Friedmann. 2001. "Connectionist models of language processing". In Smelser, Neil J.; Baltes, Paul B. (eds.). *International Encyclopedia of the Social and Behavioral Sciences*. Saint Louis: Elsevier, 2580-2584.

- Pöchtrager, Markus A. 2018. "Sawing off the branch you are sitting on." *Acta Linguistica Academica* 65(1), 47-68.
- Roach, Peter. 2009. *English phonetics and phonology: a practical course*. (4th edition). Cambridge: Cambridge University Press.
- Taylor, John R. 2002. Cognitive grammar. Oxford: Oxford University Press.
- Taylor, John R. 2006. "Where do phonemes come from? A view from the bottom." *International Journal of English Studies* 6(2), 19-54.
- Thelin, Nils. 1971. On stress assignment and vowel reduction in contemporary standard *Russian*. Uppsala: Skriv Service.
- Timberlake, Alan. 2004. A reference grammar of Russian. Cambridge: Cambridge University Press.
- Tokar, Alexander. 2017. Stress variation in English. Narr Francke Attempto: Tübingen.
- Walker, Rachel. 2011. Vowel patterns in language. Cambridge: Cambridge University Press.
- Wedel, Andrew; Ussishkin, Adam; King, Adam. 2019. "Incremental word processing influences the evolution of phonotactic patterns." *Folia Linguistica* 40(1), 231-248.
- Wells, John C. 1982. *Accents of English: an introduction. Vol. 1.* Cambridge: Cambridge University Press.

- Wells, John C. 2008. *Longman pronunciation dictionary*. (3<sup>rd</sup> edition). Essex: Pearson Education Ltd.
- Wenszky, Nóra. 2004. Secondary stress in English words. Budapest: Akadémiai Kiadó.
- Wouden, Ton van der. 1988. "CELEX: Building a Multifunctional, Polytheoretical Lexical Database". In: Magay, T. & Zigány, J. (eds.). Papers from the 3rd International EURALEX Congress. Budapest: Akadémiai Kiadó. 363–373
- Zwitserlood, Pienie. 1994. "The role of semantic transparency in the processing and representation of Dutch compounds". *Language and Cognitive Processes* 9(3), 341-368.

# 11. Appendix A

The data considered in the study is presented in Tables 9 - 11 below. The grey rows indicate the vowel phoneme. The outcome of reduction is given to the left of the words. First, the words reducing to [ $\mathfrak{p}$ ] are given, followed by those showing reduction to [ $\mathfrak{I}$ ]. The final set of words for each vowel gives examples in which no vowel reduction is found.

 Table 9 Data on short vowels

			/1/		
[1]	áktīv aktívətīj	ríðzið riðzídətij		ımplísıt ímplıkéj∫ən	místik mistísətij
	índəstrıj ındástrıjəs	místrij mistí:rijəs		lívid livídətij	prəhíbıt práwıbí∫ən
	ınsekt ınséktısajd				
[ə]	náwbəl nəwbílətrj				
			/ <del>0</del> /		
[ <del>0</del> ]	fól folfílmənt				
			/ɛ/		
[ə]	médsən mədísənəl	sɛgmént sɛgməntéj∫ən		əpélət ápəléj∫ən	
[1]	édıt ıdí∫ən	ésəns ısén∫əl		télıgra:f tılégrəfij	séləbrejt sılébrətıj
	əlέdʒ álɪgέj∫ən	pəpétʃuwɛjt pə́ːpɪtj <del>ú</del> wətɪj		mīkánīk mɛ́kəní∫ən	spəsífik spésıfísətıj
	ıkspékt ékspektéj∫ən	rīvíjl révəléj∫ən		ıkspléjn éksplənéj∫ən	prīpé: prépəréj∫ən

	tέlɪpáθɪk tɪlέpəθɪj	prızént prézəntéj∫ən	rīpé: répəréj∫ən	
[8]	métəl	əkséntrık	méntəl	féstiv
	metálık	éksentrísətıj	mentálətıj	festívətij
	ıkspékt	kəndéns	ákses	tépid
	ékspɛktéj∫ən	kóndɛnséj∫ən	əksésəbílətıj	tepídətij
	əkséntrık	əfékt	ləmént	dájəléktıks
	éksentrísətıj	áfɛktéj∫ən	lámɛntéj∫ən	dájəlektí∫ən
	səmént síjmɛntéj∫ən	o∶gmént ó∶gmɛntéj∫ən	dətést díjtɛstéj∫ən	
		/a/		
[ə]	frátənajz	válıd	ád	rápıd
	frətá:nətıj	vəlídətıj	ədíʃən	rəpídətıj
	káðlık	ávıd	ádzajl	fráðgajl
	kəðóləsajz	əvídətıj	ədzílətıj	frəðgílətrj
	mıkánık	hábıt	ásıd	dzəpán
	mékəní∫ən	həbít <del>fu</del> wəl	əsídətıj	dzápəníjz
	plásıd	fráðzajl	ákses	árıd
	pləsídətıj	frəðzílətij	əksésəbílətıj	ərídətıj
	máθəmátıks máθəmətí∫ən			
[a]	transpó:t	káptīv	áktīv	pásiv
	tránspo:téj∫ən	kaptívətīj	aktívətīj	pasívətij
	flásıd	ılástık	árəmátık	ılástık
	flasídətıj	élastísətıj	árəmatísətıj	élastísətıj

			/ɔ/	
[ə]	pózıt pəzí∫ən	kómə:s kəmá:∫əl	módən mədá:nətıj	sólıd səlídətıj
	bótənıj bətánıkəl	sajkólədzij sájkəlódziklij	pródjʉws prəd∧k∫ən	īvólv íjvəl <del>ú</del> w∫ən
	ənónəməs ánənímətıj	kómpleks kəmpléksətıj	pólıtıks pəlítıkəl	dómɛstísətıj dəmɛ́stık
	hórəbəl hərífik	hıstórık hístərísətıj	sóləm səlέmnətıj	kəndéns kóndɛnséj∫ən
	əbʻəlı∫ ábəlí∫ən	ıkónəmıj íjkənómıkəl	mórəl mərálətıj	
[0]	óltənɛjt əltə́:nətıv	hóstajl hostílətıj	flórıd flərídətıj	próspərəs prospérətıj
	tóksık təksísətıj	skwólıd skwəlídətıj		
			/ʌ/	
[ə]	káridz kəréjdzəs	s∧bstəns səbstán∫əl	kənsÁlt kónsəltéj∫ən	
[Λ]	kənfránt kónfrantéj∫ən ıgzált égzaltéj∫ən	válgə valgárətıj páblık pablísətıj	abd∧kt ábd∧ktíj	rástik rastísətij

 Table 10 Data on long monophthongs

	/1:/
[ə]	əpí: ápəríʃən

[Iː]	antí:rījə ántī:rījórətīj	díjmətíːrıjəlajz díjmətıːrıjəlajzéj∫ə	n	
		/ <del>0</del> :	/	
[ə]	bjéːrəw	əbskjó:	ədzé:	mət∫é:
	bjərókrəsıj	óbskjərέj∫ən	ádzəréj∫ən	mát∫ərɛjt
		/ε:	/	
[ə]	dəklέ:	prīpé:	dəspé:	rīpé:
	dékləréj∫ən	prépəréj∫ən	déspəréj∫ən	répəréj∫ən
	péːrənt pəréntəl			
[ɛː]	έːrɪst εːrístɪk			
		/a:	/	
[ə]	páːtıkəl	ədvá:ntıdz	sıgá:	dráːmə
	pətíkjələ	ádvəntéjdzəs	sígərét	drəmátık
[a:]	ımbá:k	ká:nəvo:	káːnəl	á:tıst
	émba:kéj∫ən	ka:nívərəs	kaːnálətıj	a:tístık
	sá:kazəm	rītá:d	stáːv	páː∫əl
	sa:kástık	ríjta:déjʃən	staːvéj∫ən	páː∫ījálətīj
	báːbərəs baːbárətıj			
		/ə:	/	
[ə]	kənfə́ːm	və́:dʒɪn	əbzə́:v	kónvə:s
	kónfəméj∫ən	vədʒínətɪj	óbzəvɛ́j∫ən	kónvəsέj∫ən
	fð:tajl	əfə́ːm	pá∶fikt	transfá:
	fətílətrj	áfəméj∫ən	pəfɛ́kʃən	tránsfərəns
	prızá∶v	rīfə:	pətéjn	prıfá:
	prézəvéj∫ən	réfəríj	páːtınəns	préfərénʃəl

	sə:kámfərəns səkámfərén∫əl	ınfə: ínfərəns	rızə́:v rézəvéj∫ən	
[əː]	á∶bən	táːbɪd	jʉwzə́:p	ıkstəːnəl
	ə∶bánətıj	təːbídətɪj	jʉwzəːpɛ́j∫ən	ékstəːnálətıj
		/o:/		
[ə]	ınfó:m	ıkspló:	ınstó:l	transfóːm
	ínfəmέj∫ən	ékspləréj∫ən	ínstəlέj∫ən	tránsfəméj∫ən
	rīstó: réstəréj∫ən	ədó: ádərέj∫ən		
[0:]	kó:z	ό:θə	ó:dıt	ɪmpóːt
	ko:zέj∫ən	o:θόrətıj	o:dí∫ən	ímpoːtέj∫ən
	kənfóːm	nóːməl	dəfóːm	fóːmat
	kónfoːméjʃən	noːmálətrj	díjfoːmέj∫ən	foːmέj∫ən
	kjo:	dəfró:d	dəpó:t	ékspo:t
	kjo:rəbílətıj	dífro:déj∫ən	díjpo:tέj∫ən	ékspo:téj∫ən
	tóːrɪst	mó:bɪd	óːgən	móːtəl
	toːrístɪk	mo:bídətɪj	oːgánık	moːtálətɪj
	pjo: pjo:rístık			

# Table 11 Data on diphthongs

	_	/ıj	/	
[1]	rīpíjt répītíʃən skíjmə skīmátīk	íjkwəl ıkwólətıj	ríjgəl rıgéjlıjə	síjkwəns sıkwén∫əl

[ə]	kəmpíjt kómpətí∫ən	əpíjl ápəlέj∫ən	rīvíjl révəléj∫ən	
[ɪj]	líjgəl lıjgálətıj	íjsθıjt ıjsθέtık		
		/	εj/	
[ə]	ıkspléjn	ıkskléjm	éjbəl	séjtən
	éksplənéj∫ən	ékskləméj∫ən	əbílətıj	sətánık
	rəwtéjt	féjtəl	stéjbəl	krəméjt
	ráwtərıj	fətálətıj	stəbílətıj	krémətóːrɪjəm
	əkléjm	gréjd	dəkléjm	ekshéjl
	ákləméj∫ən	grədéj∫ən	dékləméj∫ən	ékshəléj∫ən
	néjtrv	méjdzə	séjlajn	léjbə
	nətívətıj	mədzórətıj	səlínətıj	ləbóːrɪjəs
	prīvéjl prévələnt			
[1]	əbstéjn ábstīnəns	pətéjn pá:tınəns		
[ɛj]	néjzəl nejzálətıj	έjdʒənt εjdʒέn∫əl		
		/0	aj/	
[ə]	ədmájə	ədmájə	rıspájə	sátajə
	ádmərəbəl	ádmərέj∫ən	réspəréj∫ən	sátərajz
[1]	dəzájn	ımpláj	máltəplaj	əblájdz
	dézıgnéj∫ən	ímplıkéj∫ən	máltəplıkéj∫ən	óblıgéj∫ən
	rızájn	prızájd	rızájd	əpláj
	rézıgnéj∫ən	prézıdənt	rézıdənt	áplıkéj∫ən

	dəprájv	dərájv	dəklájn	dəfájn				
	déprīvéj∫ən	dérıvéj∫ən	déklméj∫ən	définí∫ən				
	hərájzən	əkwájə	səlájvə	fájnajt				
	hórızóntəl	ákwızí∫ən	sálıvɛjt	ínfinət				
	əpláj áplıkənt							
[aj]	fájnəl fajnálətıj	sájt sajtéj∫ən	májnə majnórətıj					
	/əw/							
[ə]	dzákstəpówz	fáwtəgra:f	páwlə	əpə́wz				
	dzákstəpəzí∫ən	fətágrəfij	pəlárətıj	ópəzí∫ən				
	dəpə́wz	ıkspówz	ımáwbajl	máwmənt				
	dépəzíʃən	ékspəzí∫ən	íməbílətıj	məméntəs				
	fáwn fənólədzıj	páwtant ímpatant	práwıbí∫ən prəhíbıt					
[əw]	īvə́wk	dənáwt	máwdəl	náwbəl				
	íjvəwkɛ́j∫ən	díjnəwtέj∫ən	məwdálətıj	nəwbílətıj				
	táwtəl təwtálətıj	rəwtéjt ráwtərıj	máwbajl məwbílətıj					
	/ʉw/							
[ə]	əkj <del>ú</del> wz	rıpj <del>ú</del> wt	rīpj <del>ú</del> wt	ımj <del>ú</del> wn				
	ákjəzέj∫ən	répjətəbəl	répjətéj∫ən	ímjənajz				
[ʉw]	br <del>ú</del> wtəl	tj <del>ú</del> wtə	kj <del>ú</del> wpıd	fl <del>ú</del> wo:				
	br <del>u</del> wtálətıj	tj <del>u</del> wtóːrɪjəl	kj <del>u</del> wpídətıj	fl <del>u</del> wórık				
	hjúwmıd	nj <del>ú</del> wtrəl	ıkskl <del>ú</del> wsıv	stj <del>ú</del> wpīd				
	hjuwmídətıj	njʉwtrálətıj	ékskl <del>u</del> wsívətıj	stj <del>u</del> wpídətīj				

	hj <del>ú</del> wmən hjʉwmánətıj	dıspj <del>ú</del> wt díspjʉwtéj∫ən	nj <del>ú</del> wtrəl nj <del>u</del> wtrálətıj	r <del>ú</del> wmətızəm r <del>u</del> wmátık			
	rıfj <del>ú</del> wt réfj <del>u</del> wtéjʃən	kəmpj <del>ú</del> wt kómpjʉwtéj∫ən					
/oj/							
[oj]	ımplój émplojíj	ıksplójt éksplojtéj∫ən	Ánəvójdəbəl Ánəvojdəbılətj				
/aw/							
[aw]	fáwnd fawndéj∫ən	áwtrejdz awtréjdzəs					

# 12. Appendix B

## 12.1. English abstract

The present thesis outlines how phonology can be dealt with from a usage-based cognitive grammar perspective by focusing on vowel reduction in Standard Southern British (SSB). In particular, the thesis proposes a way of accounting for vowel reduction using concepts such as schemas, schema interaction and categorisation relationships. The data considered in the study are taken from CUBE (Current British English searchable transcriptions) and chosen on the basis of two criteria, viz. word length (i.e. at least two syllables) and derivational morphology (i.e. a base word in which a given syllable is stressed and its derivative in which the same syllable is unstressed). Consequently, reduction processes could be observed without having to resort to underlying representations (which are, in fact, not allowed in CG). It is shown that no "ad-hoc mechanisms" (Kumashiro & Kumashiro 2006: 80) typically assumed in generative phonology and its off-shoots are needed to give a unified account of phonological phenomena. Rather, the present thesis shows that cognitive processes investigated in the field of cognitive science (e.g. schema formation) can successfully account for phonological phenomena. Additionally, the theoretical constructs developed in the analysis of English were then tested on vowel reduction in Contemporary Standard Russian (CSR). The analysis of Russian further demonstrates the promising nature of the approach.

### 12.2. German abstract

Die Studie, welche in der vorliegenden Masterarbeit vorgestellt wird, befasst sich mit einem bis heute wenig untersuchten Teilgebiet der Kognitiven Grammatik, der Phonologie. Insbesondere wird untersucht, inwieweit die gebrauchsbasierte kognitive Linguistik in der Lage ist, für Phänomene außerhalb ihrer üblichen Schwerpunkte, d.h. der Semantik und Syntax, Analysen und Lösungen aufzuzeigen. Das Phänomen, das in der Arbeit näher untersucht wird, ist die Vokalreduktion in Standard Southern British. Die Daten wurden dem Aussprachewörterbuch CUBE entnommen und anhand von zwei Kriterien ausgewählt. Da man in der Kognitiven Grammatik die Existenz von zugrundeliegenden Repräsentationen verneint, wurden Wortpaare gesammelt, die aus einem Grundwort und einem Derivativ bestehen. Ein Derivativ musste mehr als zwei Silben und einen durch Affigierung ausgelösten Betonungswechsel aufweisen. Auf diese Weise konnten Vokale in betonter und unbetonter Position direkt verglichen und Reduktionprozesse ohne Bezugnahme auf zugrundeliegendenRepräsentationen beobachtet werden. Für die Analyse der Vokalreduktion wurden kognitive Prozesse, welche für andere Teilbereiche der Linguistik (z.B. der Semantik) und der gesamten menschlichen Kognition entwickelt wurden, d.h. kognitive Schemata, Kategorisierungsbeziehungen und dergleichen, verwendet. In der vorliegenden Masterarbeit wird gezeigt, dass keine arbiträren Mechanismen nötig sind um phonologische Phänomene zu erklären. Die Grundlagen der Theorie der Kognitiven Grammatik, welche in der Semantik und Syntax zum Einsatz kommen, ermöglichen auch eine Analyse phonologischer Probleme. Abschließend umreißt die vorliegende Arbeit eine Analyse der Reduktionsphänomene im Russischen und zeigt, dass dieselben theoretischen Annahmen ohne grundlegende Abwandlungen auch für Prozesse in der russischen Sprache angewendet werden können.